UNITED NATIONS



Economic and Social Council

Distr.
GENERAL

E/CN.4/Sub.2/1999/12 28 June 1999

Original: ENGLISH

COMMISSION ON HUMAN RIGHTS
Sub-Commission on Prevention of
Discrimination and Protection
of Minorities
Fifty-first session
Item 4 of the provisional agenda

THE REALIZATION OF ECONOMIC, SOCIAL AND CULTURAL RIGHTS

The right to adequate food and to be free from hunger

<u>Updated study on the right to food, submitted by Mr. Asbjørn Eide</u>
<u>in accordance with Sub-Commission decision 1998/106</u>

CONTENTS

		<u>Paragraphs</u>	Page
I.	THE ISSUES AND THE MANDATE	1 - 12	3
II.	MALNUTRITION AND HUNGER - DIMENSIONS, CONSEQUENCES AND EMERGING ISSUES	13 - 30	5
III.	THE WORLD FOOD SUMMIT COMMITMENTS AND THEIR FOLLOW-UP	31 - 43	9
IV.	CLARIFYING THE RIGHT TO FOOD AND NUTRITION AND THE CORRESPONDING STATE OBLIGATIONS	44 - 57	13
V.	ENSURING FREEDOM FROM HUNGER AND IMPLEMENTING THE RIGHT TO ADEQUATE FOOD AT THE NATIONAL LEVEL	58 - 70	17
VI.	MONITORING AND DIALOGUE IN THE RELEVANT TREATY BODIES	71 - 84	20
VII.	THE ROLE OF THE COMMISSION ON HUMAN RIGHTS AND ITS SPECIAL RAPPORTEURS	85 - 92	23
VIII.	THE EVOLVING ROLE OF UNITED NATIONS BODIES, SPECIALIZED AGENCIES AND NON-GOVERNMENTAL ORGANIZATIONS	93 - 113	25
IX.			31
х.	CONCLUSIONS AND RECOMMENDATIONS	126 - 136	33
Annex:	General comment 12: the right to adequate food		44

I. THE ISSUES AND THE MANDATE

The issues

- 1. The Commission on Human Rights has repeatedly affirmed $\underline{1}/$ that hunger constitutes an outrage and a violation of human dignity. The present update of my previous study on the right to adequate food $\underline{2}/$ is based on the conviction that the widespread failure by States and the international community to ensure freedom from hunger and enjoyment by all of the right to food constitutes one of the most serious shortcomings of the human rights agenda. The adoption of urgent measures is required at the national, regional and international levels for the elimination of hunger and the creation of conditions in which all people can enjoy their right to food and nutrition. Unless this is done, the credibility of the human rights edifice is in serious doubt.
- 2. There is no disagreement that the right to food and nutrition is a core human right. On the occasion of the 1998 World Food Day (16 October) the President of the United States, Bill Clinton, referred to the right to food as the most basic human right. Freedom from want was one of the four fundamental freedoms referred to by an earlier United States President, Franklin D. Roosevelt, in his famous "four freedoms" address in 1941, which sparked the preparation of the Charter of the United Nations and later the Universal Declaration of Human Rights. The States parties to the International Covenant on Economic, Social and Cultural Rights have recognized that freedom from hunger is a fundamental human right, and have undertaken to take the necessary steps, to the maximum of their available resources, to eliminate hunger.
- 3. And yet more than 800 million people do not have food to meet their basic nutritional needs. This is a frighteningly high number, and hard to reconcile with the expressed commitment by States to ensure the enjoyment of human rights by everyone. There is a vast task to be completed before everyone can enjoy this fundamental human right. It is necessary, therefore, to examine the existing gap between the human rights commitments and the reality of today.

The mandate

- 4. In its decision 1997/108, the Sub-Commission requested me to review and update my study on the right to food submitted in 1987 and published in 1989. It was the first Sub-Commission study to explore the nature and content of an economic and social human right, analysing the content of the right as well as clarifying the corresponding internal and international obligations of States in regard to economic and social rights. A progress report on the update was presented at the fiftieth session of the Sub-Commission (E/CN.4/Sub.2/1998/9), which focused on major trends in the food and nutrition situation and on major events in the promotion of the right to food and nutrition. The present report should be examined together with the 1998 progress report, since information contained there will not be repeated here.
- 5. Since the completion of the 1987 study, considerable developments have taken place at the international level. The adoption of the Convention on the

Rights of the Child and its near-universal ratification considerably strengthened the place of the right to food and nutrition in international human rights law. A series of world conferences was held during the 1990s, starting with the World Summit for Children in 1990 and culminating with the World Food Summit in 1996. At nearly all of these, the problem of the extensive hunger and malnutrition in the world was addressed.

The outline

- 6. Chapter II of the present report describes the scope and consequences of malnutrition and hunger. The major purpose of that chapter is to create awareness of the seriousness of the situation. Recent scientific studies have led to the conclusion that the long-range consequences of malnutrition are much more serious than previously known.
- 7. Chapter III points out that the human rights issues are now taken seriously. Significant commitments were made at the World Food Summit in 1996, and special attention is given to the objectives set out its Plan of Action which calls for clarification of the right to food and the preparation of a coordinated strategy for its implementation. The follow-up is described, leading up to the adoption on 12 May 1999 of General Comment No. 12 by the Committee on Economic, Social and Cultural Rights (annex). Chapter IV explores the interpretation of relevant human rights provisions with particular attention to article 11 of the International Covenant on Economic, Social and Cultural Rights, drawing on General Comment No. 12.
- 8. Chapter V discusses the steps to be taken by States in order to ensure freedom from hunger and to implement the right to food. It also gives two examples of country situations where efforts to develop a human rights strategy to food have been started.
- 9. Chapter VI examines the treatment of the right to food and nutrition in relevant treaty bodies, while chapter VII reviews the treatment of the right by the Commission on Human Rights and its special rapporteurs on country situations.
- 10. Chapter VIII examines the evolving roles of the specialized agencies and other bodies in regard to the right, and shows that there has been a great increase in support for a human rights approach to food and nutrition issues.
- 11. Chapter IX briefly examines the right to food in the context of ongoing processes of globalization, and points out that these processes have created new challenges and dangers as well as new opportunities; corrective action has to be taken in order to prevent avoidable negative consequences.
- 12. Chapter X contains the conclusions and the recommendations. Some of these are addressed to States with regard to their implementation of the right to food and nutrition within their jurisdiction; others are addressed to States with regard to their international obligations towards other States and to the population living in other States. Still others are addressed to the High Commissioner for Human Rights, to the treaty bodies, and to United Nations bodies and the specialized agencies.

II. MALNUTRITION AND HUNGER - DIMENSIONS, CONSEQUENCES AND EMERGING ISSUES

- 13. The scope of malnutrition and hunger is appalling in its magnitude and an outrage to the conscience of humankind. My 1998 progress report on the update contains statistics regarding malnutrition and provides a general overview of the nature and magnitude of nutrition-related diseases. In the year that has passed since then some figures have been reconsidered and adapted, such as in the case of malnutrition assessed by underweight of children: I reported last year that around 170 million children under 5 years of age were considered underweight, while the most recent authoritative figure is put at "more than 200 million". $\underline{3}/$
- 14. The Director-General of the World Health Organization recently gave the following examples of the impact of malnutrition on disease: $\frac{4}{}$ 208 million people are stunted and 49 million are wasted; more than 900 million people suffer from goitre, 16 million are severely retarded and another 50 million suffer other forms of brain damage due to a deficiency of iodine; 3 million children are at increased risk of infection, blindness and death because they are vitamin A-deficient; anaemia and iron deficiency affect more than 2,000 million people the world over; and some 22 million children and more than 200 million adults are obese, and thus at significant risk from a range of serious non-communicable diseases and other threats to health.
- 15. One can hardly say that the overall problem has improved much although there are some positive signs and trends in some parts of the world (see the 1998 progress report), and there are single-country experiences that may give hope to other States wishing to address the problem in a sustained manner. $\underline{5}/$
- 16. This final report highlights some of the most significant major new concerns and ways of understanding the causes of malnutrition as they have emerged since my study was submitted in 1987. Scientific evidence and practical experience over the last decade have converged on a much more integrated picture than we were able to draw in 1987. Only by understanding the nature and complexity of the problems stemming from the non-fulfilment of the right to adequate food, can we interpret this right more fully and identify the corresponding national and international obligations of States.
- 17. Inadequate access to food and suboptimal nutritional status may have multiple interacting causes, some of them deeply rooted in malfunctioning of the society. The previously widespread but erroneous notion that insufficient global and even national food production was the cause of hunger has now generally been discarded. There is greater recognition of the need for systemic and integrated analyses and approaches towards solutions. The translation from theoretical principles to practical action requires transcending the borders of single sectors and organizations with their often firmly established agendas and portfolios.
- 18. Three areas need particular attention: firstly, the new emphasis on the so-called "life-cycle approach" to understanding and addressing malnutrition; secondly, the revitalized associated understanding of the critical role of women and young girls as mediators of both societal and biological processes

leading to and perpetuating malnutrition between generations; thirdly, the rapidly changing distribution of the "global burden of disease" between the developing and the industrialized world as well as within the developing world. In addressing all of these concerns human rights ought to come into play, enforcing consideration of the "right to adequate food" in a very broad economic, social and human development perspective. $\underline{6}/$

The "life-cycle approach" to understanding malnutrition

- 19. The pernicious impact of malnutrition can affect the whole life-span of human beings. This has now become obvious and makes it even more urgent to take wide-ranging corrective measures. We know that poverty is a vicious circle and malnutrition may greatly affect work capacity in adults, thus contributing to the perpetuation of poverty. The tremendous impact of malnutrition on human, family and societal development is indisputable.
- 20. Until recently the effects of hunger and various forms of malnutrition were normally related to susceptibility at various age-levels and remedies were considered in terms of the need to attack the problem at that stage through food supplementation or other forms of rehabilitation. Although there was always a discussion of the consequences of infant and young child malnutrition in terms of small, but otherwise healthy adults, as well as a certain undocumented fear of permanent brain damage from foetal undernutrition, it was only in the 1990s that new theories and evidence from long-term studies led to the idea that early-life malnutrition also affected susceptibility to disease in later life.
- 21. Findings and theories put forward during the present decade suggest that foetal and young-infant malnutrition may in fact "programme" an individual for susceptibility to nutrition-related chronic diseases in later life, such as heart diseases, obesity and diabetes. It is ironic that these diseases, which were thought to be caused by "over-eating" in the industrialized world, are now sweeping the poor world with unprecedented speed. They imply disability, disease and death much earlier than expected for people who survived the critical years of childhood in poor societies. Furthermore, the risk factors which add to the biological programming effects are themselves frequently related to poverty, for example, certain fat and sugary foods are known to produce "cheap energy". 7/
- 22. The scope and gravity of this life-cycle malnutrition is closely related to adverse social and biological factors which may combine to influence the nutrition and reproductive performance of women in ways that may critically affect their offspring while still in the womb and have consequences for the life-cycle of the next generation. The ACC/SCN report summarizes the seriousness of the complex interrelationships at work as follows:

"The acceptance by the international community that poorly fed adults are economically less effective now needs to be linked to new findings of up to 1 billion adults who are malnourished, with a reduced work capacity and an enhanced susceptibility to infections. The remarkably high rates of anaemia in children, men and women in many parts of the globe not only impair work capacity but permanently damage the normal brain development of infants. Now, however, it is also clear

that the commonest form of 'childhood' malnutrition, involving stunted growth, is also closely linked to impaired mental as well as physical development. Thus, the capacity of societies to develop economically is being handicapped for decades by our current failure to eliminate malnutrition and anaemia in both children and adults. The impact of poor nutrition is therefore now seen as affecting the whole life-cycle. Malnourished mothers, particularly in Asia, are producing low-birth-weight babies who are not only immediately at risk but also become stunted as young children, with a double handicap of physical and mental limitations. In some societies this affects half of the population, with new evidence that in Asia poorly growing girls grow into malnourished adults who then in pregnancy pass on the long-term impact of their own poor development during foetal life to the next generation. To break this intergenerational cycle is exceptionally important - not only a fundamental ethical issue but also a priority for any Government concerned for the future intellectual and economic capacity of its people." (emphasis added)

The particular role of women and young girls as both victims and "mediators" in the generation of malnutrition

- 23. Women who are socially disadvantaged in ways that can affect their reproductive functions can contribute to a vicious circle: young girls are often underfed and mothers-to-be may start their early pregnancies in a nutritionally suboptimal state, giving birth to low-birth-weight children who may also have had suboptimal nourishment in the womb. Added to such a bad start in life may be inadequate feeding patterns and repeated infectious diseases in infancy and as a young child. These adverse factors may combine to affect vital human development processes, including brain development, in the most critical formative years.
- 24. Underfed and under-cared-for mothers are thus affected socially and biologically, demonstrating the need for a continuing gender-role analysis in all undertakings aimed at improving food and nutrition. This must in turn be seen in the context of responsibilities and rights of women, including those of young girls. We are faced with both an ethical imperative vis-à-vis women themselves, and an imperative to safeguard future generations from being exposed to the same mutually reinforcing risks and thereby further perpetuating the situation. The vicious circle must be broken.
- 25. The problem has been that women's chores giving birth and nurturing the new generation, as well as seeing to household food and economic security overall, have too often been considered as belonging to the domestic sphere and therefore outside State responsibility. We have begun to understand better to what extent these chores are at the very heart of problems that, because they are often defined in aggregate terms, tend to be missed when identifying crucial points for intervention. Renewed consideration of the linkages between women's rights and the right to adequate food and feeding is warranted, in the context of the consideration of the systemic linkages between the processes that create poverty and the manifestation of poverty itself, and cultural factors resulting in discrimination against women.

26. A special case in point is women's opportunities for breastfeeding their children, thereby providing them with the most appropriate food for infants at least up to 6 months of age - human milk. The State has an obligation to respect women's right to provide this food and to protect and facilitate the necessary conditions for them to do so, at least for six months. This may require legislative and supportive measures protecting mothers from being forced to stop nursing or to shift to alternative food, including mothers' milk substitutes, before their infants' physiology has matured to a stage when they can tolerate and utilize food other than human milk.

The distribution of the burden of nutrition-related disease

- 27. It is also clear from what has been said about the life-cycle approach that the picture we used to have of hunger, malnutrition and ill-health being caused by inadequate food intake undernutrition in developing countries and "overnutrition" in the rich countries has changed, and dramatically so. We find all the major nutrition-related diseases in most countries, and spread among social classes at that. This has led to the expression "the double burden of disease" to describe what poor societies now are experiencing. The typical diseases found in developing societies above all infectious diseases which in combination with undernutrition lead to high mortality, especially among children continue to take their toll and drain poor health budgets; in addition, the new diseases add to the drain in terms of early disability or loss of breadwinners. 8/
- 28. Especially intriguing is the finding of differential rates of malnutrition among women and young children in different developing societies. The most outstanding example is provided by a comparison of trends in South Asia and in sub-Saharan Africa. Women and young children in South Asia appear to be significantly worse off than in Africa. Several theories have been put forward to try to explain this without any firm conclusions having been reached. The lesson to be drawn, however, in assessing the nutritional status of populations is that the underlying causality is often much more complex than a single-factor explanation such as overall lack of food. And yet, adverse nutritional trends are all too often used to "demonstrate" the need for increased food production as the remedy. The "right to adequate food" may be as much a question of the full realization of the rights of women as of ensuring a bundle of nutrients handed over through food supplementation schemes.
- 29. To combat malnutrition is required not only from a human rights perspective; the case for nutrition is a strong one also in economic terms. James Lovelace, Director of the Health, Nutrition and Population Network of the World Bank, argues that development thinkers, planners and practitioners have for too long failed to recognize how, and how deeply, malnutrition is implicated in the genesis and persistence of poverty. "The pervasive malnutrition of children, and also of women and men, has surreptitiously undermined and frustrated development efforts in many fields. Nutrition should become the new frontier, attacking stunted development and compromised growth of people, especially the young." He also argued that we know how to solve the problem of malnutrition: "Carefully selected interventions and

approaches are not only a human rights imperative but make eminent economic sense as well. States and the international community cannot afford not to do so, because malnutrition is very expensive for society." $\underline{9}/$

30. Human rights arguments and economic arguments in favour of improved food and nutrition are thus both important and can live side-by-side, not being opposed to one another. The point is that the rationale for human beings to be well fed should not be perceived and advocated in economic terms alone - the dignity of men, women and children must be superimposed on any mere utilitarian argument. Human rights pays off economically, and successful economic interventions will enhance the realization of human rights: these are two sides of the same coin and both can be promoted.

III. THE WORLD FOOD SUMMIT COMMITMENTS AND THEIR FOLLOW-UP

- Notwithstanding the fact that the Universal Declaration of Human Rights places all human rights on the same level, the attention given at the international level to economic and social rights was appallingly low for several decades. One of the purposes of my 1987 study was to improve the understanding of economic and social rights and through that to advance their recognition and implementation. Significant advancement in the recognition and promotion of these rights was made through a series of important world conferences: the World Summit for Children in 1990, the United Nations Conference on Environment and Development (the Rio Conference) in 1992, the International Conference on Nutrition in 1992, the World Conference on Human Rights in 1993, the International Conference on Population and Development in 1994, the World Summit for Social Development in 1995, the Fourth World Conference on Women in 1995 and the second United Nations Conference on Human Settlements (Habitat II) in 1996. They constituted steps in a process towards a better understanding of the relationship between human rights and development, and a recognition that economic and social rights are as indispensable as civil and political rights. The contribution by each one of these conferences to economic and social rights has been outlined in my 1998 progress report (paras. 22-29), to which the reader is referred.
- 32. The major breakthrough with regard to the right to food came with the World Food Summit, held at the invitation of the Food and Agriculture Organization of the United Nations (FAO) in Rome in November 1996. Taking into account that the Rome Declaration on Food Security and World Food Summit Plan of Action was adopted by the Summit and thus by the leaders of the world, the commitments made there must be considered to have overriding legal, political and moral significance for the world community.
- 33. The Governments assembled in Rome declared that they considered it intolerable and unacceptable that more than 800 million people throughout the world, particularly in developing countries, did not have enough food to meet their basic nutritional needs. They recognized that this was not a result of lack of food supplies. They noted that such supplies had increased substantially. The constraints were to be found in the access to food. The problems, they noted, resided in the continuing inadequacy of household incomes to purchase food and the instability of supply and demand, as well as natural and man-made disasters.

- 34. They also pointed out that the problems of hunger and food insecurity had global dimensions and were likely to persist and even increase dramatically in some regions unless urgent, determined and concerted action was taken, especially given the anticipated increase in the world's population and the stress on natural resources. The leaders of the world therefore pledged their political will and their common and national commitment to achieving food security for all and to an ongoing effort to eradicate hunger in all countries, with an immediate view to reducing the number of undernourished people to half the 1996 level no later than the year 2015.
- 35. To implement this political will, the leaders of the world made the following seven commitments which were spelled out in the Plan of Action:
 - 1. To ensure an enabling political, social and economic environment designed to create the best conditions for the eradication of poverty and for durable peace, based on full and equal participation of women and men, which is most conducive to achieving sustainable food security for all;
 - 2. To implement policies aimed at eradicating poverty and inequality and improving physical and economic access by all, at all times, to sufficient, nutritionally adequate and safe food and its effective utilization;
 - 3. To pursue participatory and sustainable food, agriculture, fisheries, forestry and rural development policies and practices in high and low potential areas, which are essential to adequate and reliable food supplies at the household, national, regional and global levels, and combat pests, drought and desertification, considering the multifunctional character of agriculture;
 - 4. To strive to ensure that food, agricultural trade and overall trade policies are conducive to fostering food security for all through a fair and market-oriented world trade system;
 - To endeavour to prevent and be prepared for natural disasters and man-made emergencies and to meet transitory and emergency food requirements in ways that encourage recovery, rehabilitation, development and a capacity to satisfy future needs;
 - 6. To promote optimal allocation and use of public and private investments to foster human resources, sustainable food, agriculture, fisheries and forestry systems, and rural development, in high and low potential areas;
 - 7. To implement, monitor, and follow-up the Plan of Action at all levels in cooperation with the international community.

- 36. Commitment 7 (implementation, monitoring, and follow-up) is of special importance for the present study. The following objectives in the pursuit of commitment 7 were specified:
 - 7.1 To adopt actions within each country's national framework to enhance food security and enable the implementation of the commitments of the World Food Summit Plan of Action;
 - 7.2 To improve subregional, regional, and international cooperation and to mobilize, and optimize the use of, available resources to support national efforts for the earliest possible achievement of sustainable world food security;
 - 7.3 To monitor actively the implementation of the World Food Summit Plan of Action;
 - 7.4 To clarify the content of the right to adequate food and the fundamental right of everyone to be free from hunger, as stated in the International Covenant on Economic, Social and Cultural Rights and other relevant international and regional instruments, and to give particular attention to implementation and full and progressive realization of this right as a means of achieving food security for all;
 - 7.5 To share responsibilities in achieving food security for all so that implementation of the World Food Summit Plan of Action takes place at the lowest possible level at which its purpose could be best achieved.
- 37. The right to food is thus directly addressed in objective 7.4. The Plan of Action called for the following steps to be taken by Governments, in partnership with all actors of civil society:
- (a) Make every effort to implement the provisions of article 11 of the International Covenant on Economic, Social and Cultural Rights and relevant provisions of other international and regional instruments;
- (b) Urge States that are not yet parties to the Covenant to adhere to the Covenant at the earliest possible time;
- (c) Invite the Committee on Economic, Social and Cultural Rights to give particular attention to this Plan of Action in the framework of its activities and to continue to monitor the implementation of the specific measures provided for in article 11 of the Covenant;
- (d) Invite relevant treaty bodies and appropriate specialized agencies of the United Nations to consider how they might contribute, within the framework of the coordinated follow-up by the United Nations system to the major international United Nations conferences and summits, including the World Conference on Human Rights, within the scope of their mandates, to the further implementation of this right;

(e) Invite the United Nations High Commissioner for Human Rights, in consultation with relevant treaty bodies, and in collaboration with relevant specialized agencies and programmes of the United Nations system and appropriate intergovernmental mechanisms, to better define the rights related to food in article 11 of the Covenant and to propose ways to implement and realize these rights as a means of achieving the commitments and objectives of the World Food Summit, taking into account the possibility of formulating voluntary guidelines for food security for all.

The follow-up

- 38. The High Commissioner for Human Rights has taken several steps to initiate cooperation with the relevant agencies and treaty bodies. A memorandum of understanding with FAO was adopted for the implementation of objective 7.4. The Committee on Economic, Social and Cultural Rights held a Day of General Discussion on the right to food, attended by the High Commissioner and invited experts, on 1 December 1997, followed by the first expert consultation on the matter on 2 December. The main conclusions of that consultation are included in my 1998 progress report (paras. 36-37). They included a recommendation to the Committee on Economic, Social and Cultural Rights to prepare and adopt a general comment on the right to food, and steps were set in motion for this. In November 1998 a second consultation was organized by the Office of the High Commissioner for Human Rights and co-hosted by FAO in Rome. The conclusions of this consultation were reported by the High Commissioner to the Commission on Human Rights in April 1999.
- The recommendation of the World Food Summit to consider the preparation of "voluntary guidelines" was carried forward in 1997 by three non-governmental organizations or institutions - FIAN - Foodfirst Information and Action Network, the World Alliance on Nutrition and Human Rights and the Jacques Maritain Institute. They started early in 1997 to draft a code of conduct on the human right to adequate food which went through a thorough process of NGO hearings and amendments until a final draft was issued in September 1997. The draft code has already been endorsed by more than 800 NGOs. It is intended to be brought before the relevant international bodies for their consideration at an appropriate time. The draft code of conduct is a well-crafted document responding to the two components of the call by World Food Summit: to clarify the content of the right to adequate food and the fundamental right of everyone to be free from hunger, and to give particular attention to implementation and full and progressive realization of this right as a means of achieving food security for all. Further details about the content of the code is found in chapter VIII of this report.
- 40. The body that initiated the preparations of the World Food Summit negotiations and whose major task it now is to coordinate the follow-up of the Summit's Plan of Action the FAO Committee on Food Security has so far kept a lower profile vis-à-vis objective 7.4 follow-up than might have been expected, possibly waiting for a comprehensive response to the mandate given to the High Commissioner about content clarification and a strategy for implementation of a human rights approach to food security.
- 41. As to the technical food and nutrition agencies and their staff, the internalization of the value of a human rights approach to food and nutrition

assessment, advocacy and programming has slowly but steadily matured. Within the ACC/SCN, the working group on Nutrition Ethics, and Human Rights has for some years been examining the meaning of a human rights approach to food and nutrition within the scope of development assistance. At its recommendation the ACC/SCN devoted its annual-session symposium in 1999 to "The Substance and Politics of a Human Rights Approach to Food and Nutrition Policies and Programmes", which was held in Geneva on 12 and 13 April 1999 and hosted by the High Commissioner for Human Rights. The symposium was attended by senior officials of several agencies together with their technical focal points, proof of an increasing receptivity to human rights among the development agencies. And while a few agencies have for some time been in the front line in advancing a human rights approach (notably UNICEF, more lately FAO), the symposium contributed to consolidating a necessary change in the attitudes of others and a better understanding of the work of the human rights institutions as well as their own place in future collaboration with them. Chapter VIII of this report summarizes the positions taken and the commitments made by representatives of many of the agencies attending this important symposium.

- 42. The most significant milestone to date in the follow-up process is the preparation and adoption, by the Committee on Economic, Social and Cultural Rights at its twentieth session, of General Comment No. 12 (E/C.12/1999/5) on the right to adequate food (art. 11) on 12 May 1999. That general comment is of central importance and is therefore annexed to this report. Its contents are considered in chapter IV.
- 43. The net result of these activities since the World Food Summit is that the content of the right to food and nutrition has been made more clear, the awareness of its significance has become more widespread, and more resolute steps are now being taken to implement this right.

IV. CLARIFYING THE RIGHT TO FOOD AND NUTRITION AND THE CORRESPONDING STATE OBLIGATIONS

- The right to food forms part of the broader right to an adequate standard of living. $\underline{10}$ / The right to an adequate standard of living - or to livelihood - sums up the main concern underlying all economic and social rights, which is to integrate everyone into a humane society. The ultimate purpose of promoting the right to adequate food is to achieve nutritional well-being for the individual child, woman and man. Human nutritional status is determined by at least three major clusters of conditions which interact in a dynamic fashion, relating to food, health and care, and with education as a cross-cutting dimension. Food alone is not sufficient to ensure good nutrition for the individual. The right to adequate food is a necessary, but not alone sufficient component of the right to adequate nutrition. The full realization of the latter depends also on parallel achievements in the fields of health, care for the vulnerable, and education. Later instruments, especially the Convention on the Rights of the Child, recognize this broader right to nutrition as well as its cross-cutting nature relative to many other rights, including civil and political rights.
- 45. Article 11 of the ICESCR focuses on the food dimensions of the broader concept. The call by the World Food Summit for a better definition of the right to food as contained in article 11 has now in broad outline been met

through the process described in chapter III and culminated with the adoption by the Committee on Economic, Social and Cultural Rights of General Comment No. 12. This is a most significant step in the process towards clarification of the content of the right to food and the steps that should be taken for its realization. Given its origin as the official interpretation by the treaty body responsible for monitoring States parties' implementation of the right to adequate food, this general comment will in the time ahead stand as the most authoritative document formulated to date regarding the right to food.

- 46. The full text of the general comment on the right to adequate food should be studied as an integral part of this update. The author fully endorses the content of the general comment. Below follow some observations regarding the structure and the main highlights of its contents: the dignity, adequacy and sustainability of availability and access to food.
- 47. The foundation principle of the human rights system is that everyone is "born free and equal in dignity and rights and should act towards each other in a spirit of fraternity" (UDHR, art. 1). The Committee affirms in its general comment (para. 4) that "the right to adequate food is indivisibly linked to the inherent dignity of the human person and is indispensable for the fulfilment of other human rights enshrined in the International Bill of Human Rights. It is also inseparable from social justice, requiring the adoption of appropriate economic, environmental and social policies, at both the national and international levels, oriented to the eradication of poverty and the fulfilment of all human rights for all."
- 48. The general comment expresses the core content of the right to adequate food as follows (para. 8):
 - The availability of food in a quantity and quality sufficient to satisfy the dietary needs of individuals, free from adverse substances, and acceptable within a given culture;
 - The accessibility of such food in ways that are sustainable and that do not interfere with the enjoyment of other human rights.
- 49. The content of "adequacy" is to some extent determined by prevailing social, economic, cultural, climatic, ecological and other conditions, but there are some core elements which are applicable under all circumstances and are spelled out in greater detail by the Committee. "Sustainability" incorporates the notion of long-term availability and accessibility and is intrinsically linked to the notion of adequate food or food security, implying food being accessible for both present and future generations.
- 50. In clarifying the content of the term adequate, the general comment (paras. 7-11) lays out in some detail the meaning of the sub-concepts of dietary needs, absence of adverse substances, and cultural or consumer acceptability. The concept of sustainability is explained in terms of availability and accessibility both physical and economic for different groups of people (paras. 12-13). This level of detail is very useful as it offers an opportunity for more nuanced reporting by States parties on the realization of the right to adequate food, which only attains full meaning

when "food" also reflects the enjoyment of other rights which will ultimately determine accessibility and constitutes a true carrier of nutritional health.

On State obligations, nationally and internationally

- 51. In very general terms, the legal obligations of States parties to the ICESCR which are set out in article 2 of the Covenant have been spelled out in earlier general comments and are now spelled out in paragraphs 14-16 of the general comment, to which I refer.
- 52. My 1987 study was intended as a contribution to the clarification of the nature and levels of State obligations under economic and social rights. Drawing on my earlier work, I introduced an analytical framework under which State obligations can be assessed on three levels: the obligation to respect, the obligation to protect, and the obligation to assist and fulfil human rights. The framework proved very useful, and has since been taken widely into use. In my progress report in 1998 I explained my previously elaborated framework in greater detail, as follows: (para. 9).
- (a) Since State obligations must be seen in the light of the assumption that human beings, families or wider groups seek to find their own solutions to their needs, States should, at the primary level, respect the resources owned by the individual, her or his freedom to find a job of preference, to make optimal use of her/his own knowledge and the freedom to take the necessary actions and use the necessary resources alone or in association with others to satisfy his or her own needs. The State cannot, however, passively leave it at that. Third parties are likely to interfere negatively with the possibilities that individuals or groups otherwise might have had to solve their own needs;
- (b) At a secondary level, therefore, State obligations require active protection against other, more assertive or aggressive subjects more powerful economic interests, such as protection against fraud, against unethical behaviour in trade and contractual relations, against the marketing and dumping of hazardous or dangerous products. This protective function of the State is widely used and is the most important aspect of State obligations with regard to economic, social and cultural rights, similar to the role of the State as protector of civil and political rights;
- (c) At the tertiary level, the State has the obligation to facilitate opportunities by which the rights listed can be enjoyed. It takes many forms, some of which are spelled out in the relevant instruments. For example, with regard to the right to food, the State shall, under the International Covenant (art. 11 (2)), take steps to "improve measures of production, conservation and distribution of food by making full use of technical and scientific knowledge and by developing or reforming agrarian systems".
- (d) At the fourth and final level, the State has the obligation to fulfil the rights of those who otherwise cannot enjoy their economic, social and cultural rights. This fourth level obligation increases in importance with increasing rates of urbanization and the decline of group or family responsibilities. Obligations towards the elderly and disabled, which in

traditional agricultural society were taken care of by the family, must increasingly be borne by the State and thus by the national society as a whole.

- 53. The Committee has now generally endorsed this approach in its general comment 12 (para. 15). It will be noted that the Committee has decided to use the three levels of obligations, as originally suggested by me, but has subdivided the third level into the obligation to facilitate and to provide (fulfil), as I have suggested in my later work.
- 54. Economic and social rights, like all other human rights, are often violated, but there has been some controversy over what constitute a violation of such rights, including the right to food. 11/ This is dealt with in the general comment (paras. 17-20). Violations of the Covenant occur when a State fails to ensure the satisfaction of, at the very least, the minimum essential level required to be free from hunger. Resource constraints do not free the Government from responsibility; it still has to show that every effort has been made to use all the resources at its disposal in an effort to satisfy, as a matter of priority, those minimum obligations. Any discrimination in access to food, as well as to means and entitlements for its procurement, constitutes a violation of the Covenant. Violations can occur through the direct action of States or other entities insufficiently regulated by States. Non-State actors also have responsibilities, and the State should provide an environment that facilitates implementation of these responsibilities. In regard to the private business sector, whether national or transnational, a code of conduct should be agreed upon and made use of.
- 55. While article 11 of the ICESCR is the key provision in regard to the right to food, it should be kept in mind that issues relating to food and nutrition are dealt with in numerous other international instruments. Of particular importance are article 24 of the Convention on the Rights of the Child and articles 12 and 14 of the Convention on the Elimination of All Forms of Discrimination against Women, both of which deal with the right to adequate nutrition.
- 56. Important provisions are also found in international humanitarian law as laid down principally in the four Geneva Conventions of 1949 and the two Additional Protocols of 1977. It prohibits the starvation of civilian populations, as well as the destruction of objects indispensable to their survival, such as foodstuffs, agricultural areas, crops, livestock, drinking water installations and irrigation works. It also prohibits methods or means of warfare likely to cause widespread, long-term damage to the environment, thereby jeopardizing the health or survival of the population. It contains provisions concerning humanitarian assistance and relief operations, including the free passage of such essentials as food, medicines and other goods of primary necessity.
- 57. Much of international humanitarian law was designed for an environment of armed conflict which has undergone significant changes in recent years, notably since the end of the cold war. There has been a shift from conflicts between States the traditional focus of humanitarian law to armed conflicts within States, and a rise in intra-State tensions and disturbances short of outright armed conflict. Human rights, including freedom from

hunger, continue generally to apply also during armed conflict. The allocation of responsibility between the different actors involved in the conflict, as well as the rights, obligations and roles of international agencies, are matters of controversy in great need of clarification.

- V. ENSURING FREEDOM FROM HUNGER AND IMPLEMENTING THE RIGHT TO ADEQUATE FOOD AT THE NATIONAL LEVEL
- 58. Conditions differ widely between countries and the most appropriate ways and means to ensure freedom from hunger and to implement the right to adequate food therefore also varies considerably. Each State party to the ICESCR and the Convention on the Rights of the Child must nevertheless take the measures required under the particular conditions prevailing in its country to ensure that everyone is free from hunger, and thereupon move as quickly as possible to a situation where everyone can enjoy their right to adequate food. In so doing, the different levels of State obligations outlined in the general comment and discussed above in chapter IV should be applied, using the particular combination of measures warranted by the national situation.
- 59. Every country should therefore develop a national strategy to implement the right to food, or if such a national strategy is already in place, it should be revised in light of the observations made in general comment 12. The first step in any such strategy should be to map the situation for the different groups and the different regions within the country, taking into account the differences that might exist on the basis of gender, ethnicity or race, and between rural and urban areas. Such mapping is necessary to identify those who are food insecure and to develop appropriate responses to food insecurity. $\underline{12}/$
- 60. Food security is currently defined as access by all people at all times to the food needed for a healthy and active life. 13/ Achieving food security means ensuring that sufficient food is available, that supplies are relatively stable and that those in need of food can obtain it. 14/ National food security would therefore mean that everyone has effective entitlements to adequate food or resources for food and that there is in principle enough food to go around. Achieving food security at household level means ensuring that sufficient food is available throughout the territory, that supplies are relatively stable and that everyone within that territory in need of food has the capacity to obtain it for a healthy and productive life. Food insecurity is understood as the reverse: where some people at some or all times do not have access to adequate food and therefore cannot lead a healthy and active life.
- 61. Food security can therefore be seen as the corollary of the right to food: the State must take the necessary measures to ensure food security for everyone under its jurisdiction. In their national efforts to identify the groups which are food insecure, States should review periodically and/or when special circumstances cause significant changes, including serious economic depressions who is likely to be affected and how different groups may be affected differently on the basis of chronic or transitory vulnerability.

- 62. Having identified the groups which are food insecure, the national strategy should recreate food security for those groups and the population as a whole, formulate policies required for this purpose and identify the resources available, including human resources which, through self-help, can contribute to meet the goals set. When new and severe resource constraints emerge, caused by a process of economic adjustment, economic recession, climatic conditions or other factors, measures should be undertaken to ensure as a minimum that vulnerable population groups and individuals do not face hunger. This should be achieved through social programmes, safety nets and international assistance.
- 63. The right to food can only be fully achieved if it is linked to the realization of all other human rights. In the elaboration and implementation of the national strategy for the right to food, people's informed participation is essential. Appropriate decentralization is required to ensure that the particular conditions in the different regions of the country are taken into account, but the central Government remains at all times ultimately responsible for ensuring the enjoyment of this and other human rights.
- 64. Care must be taken to ensure sustainable management of natural and other resources required for access to food, including the management of land, water and fishery resources. Furthermore, attention must be given to all aspects of the food chain, from production to consumption. Attention must further be given to the processing of food in order to ensure food safety, to the physical distribution of food in order to enable access for all, and to improving markets, inter alia in order to prevent misleading and potentially harmful influences. At the level of the consumer, adequate information is required on the composition of healthy diets. Sanitary conditions and clean water are essential in ensuring that the food actually consumed is safe.
- The strategy should give particular attention to the need to prevent discrimination in access to food or resources for food. It should keep in mind the need to give particular attention to vulnerable groups and to the gender situation. Indigenous peoples and some minorities often suffer from more serious malnutrition than the average population, and we have seen that women in many societies face far greater difficulties than men. The strategy should therefore include, inter alia: guarantees of full and equal access, particularly for women, to economic resources, including the right to inheritance and ownership of land and other property, credit, natural resources and appropriate technology; measures to respect and protect self-employment and work which provides a remuneration ensuring a decent living for wage earners and their families (as stipulated in article 7 (a) (ii) of the Covenant); maintaining registries on land rights (including forests) and fishing resources important for food production or other means of livelihood in rural areas; special legislation to protect the land rights of indigenous peoples and prevent forced evictions from their land; measures to protect or secure entitlements to land or other sources of livelihood among vulnerable smallholders, landless peasants and urban poor without discrimination as to race or ethnicity.

Strategy elaboration: Brazil and South Africa

- 66. Several positive examples exist of States where efforts are now being made to elaborate a human rights-based strategy to ensure freedom from hunger and the enjoyment of the right to food. Brief mention will be made here of two countries which, each in its own way, have gone through great political difficulties and where severe social inequalities exist, but where steps are now under elaboration to address in a direct way the human rights aspects of food and nutrition.
- 67. Brazil traditionally has been socially divided due to severe differences in income, which partly explains past occurrences of military rule. Democratic efforts towards social integration and greater equality have given rise to efforts to develop a strategy for the elimination of hunger and the right to food. The National Food Security Council, established in 1993, was replaced in 1995 by the broader State-civil society link mechanism Communidade Solidária, which operates within the Presidency. The National Human Rights Programme, which was enacted in 1996, focused initially solely on civil and political rights. The food security movement, in its preparations for the World Food Summit, recognized that a human rights framework was required. This resulted in a partial merging of the human rights movement and the networks devoted to food security and nutrition.
- 68. A strategy-building cooperative network now involves, inter alia, the Ministry of Health, the Ministry of Planning, UNICEF, and the non-governmental network Agora. In 1998 the Ministry of Health initiated a process of developing health policies. Following discussions with representatives of the civil society it was agreed to make use of the human rights approach to health. The first sector chosen for policy development and implementation was the national food and nutrition policy. A step-by-step process has been initiated. The first area of concern was iron deficiency anaemia, which is said to be the most prevalent nutrition problem in Brazil. A task of a different order has been to train nutrition managers at different levels to take into account the human rights approach to their work. 15/
- 69. In South Africa, apartheid was a policy deliberately intended to create and maintain differences. Its legacies are not easily overcome in post-apartheid South Africa. Social indicators show an extreme difference in income and consumption between the rich (most of whom are White) and the poor (most of whom are Africans). The South African Constitution, adopted in 1996, is intended to transform the society so that everyone can enjoy their freedom and dignity. The Constitution ranks as one of the best in the world with its inclusion of all human rights, the economic, social and cultural as well as the civil and political. In the Bill of Rights, section 7 (2), the State is required to respect, protect, promote and fulfil the realization of all human rights, including social and economic rights.
- 70. The South African Human Rights Commission has, under section 184 (1) of the Constitution, been given the task, <u>inter alia</u>, to monitor and assess the observance of human rights in South Africa. This gives the Commission powers to investigate and report on the observance of human rights; take steps to secure appropriate redress where human rights have been violated; conduct research on the realization of human rights and provide human rights

education. From 25 to 27 March 1999 the Commission, in cooperation with UNICEF and the World Alliance for Nutrition and Human Rights, held a conference in Johannesburg on the right to food and nutrition (to which this author was invited as a speaker). It can be seen as a first step towards the development of a national strategy to ensure freedom from hunger and the right to food. Extensive participation by representatives of government departments and representatives of the civil society gave rise to a set of constructive proposals to further develop a human rights-oriented national food and nutrition strategy. $\underline{16}/$

VI. MONITORING AND DIALOGUE IN THE RELEVANT TREATY BODIES

71. International monitoring of human rights is essential for effective and comprehensive implementation. The process of monitoring provides opportunities for dialogue and learning. Since three of the main international human rights instruments deal with the rights to food and nutrition, three treaty bodies are involved in that dialogue. This chapter provides some examples of food and nutrition issues which have been dealt with in these bodies. The cases are brought up solely as illustrations; others could have been chosen but are not included here for lack of space.

Committee on Economic, Social and Cultural Rights

- 72. The Committee on Economic, Social and Cultural Rights is the main treaty body addressing the right to food. With the adoption in May 1999 of its general comment 12 on the right to food it now has a solid basis for its monitoring of national implementation of that right, particularly if States parties follow the recommendation to adopt a strategy, including framework legislation and the establishment of verifiable benchmarks for subsequent national and international monitoring. 17/
- 73. Few States parties have in their reports provided information sufficient and precise enough to enable the Committee to determine the prevailing situation in the countries concerned and identify the obstacles to the realization of the rights under discussion. Provided States parties cooperate in the future along the lines suggested in general comment 12, it will become much easier to monitor future developments regarding the right to food.
- 74. Country situations where the Committee has expressed serious concerns over violations of the right to food include Nigeria and Guatemala. In the concluding observations in the case of Nigeria (E/C.12/1/Add.23 of 13 May 1998), the Committee stated that it was greatly disturbed that 21 per cent of the population of Nigeria was living below the poverty line in spite of the country's rich natural resources. The Committee further noted with concern that due to poor economic and administrative mismanagement, corruption, runaway inflation and the rapid devaluation of the naira, Nigeria ranked among the world's 20 poorest countries. The Committee pointed out that according to World Bank estimates at least 17 million Nigerians, many of them children, were undernourished.
- 75. In its concluding observations regarding Guatemala (E/C.12/1/Add.3 of 28 May 1996) the Committee expressed serious concern at the adverse effects of the economic and social disparities on the enjoyment of economic, social

and cultural rights by the majority of the population, particularly by the indigenous and rural populations of Guatemala. The Committee noted that far-reaching racial discrimination, extreme poverty and social exclusion in relation to the indigenous populations negatively affected their enjoyment of economic, social and cultural rights, and were matters of deep concern for the Committee. While the Committee appreciated the open admission of the Government that land had been illegally appropriated by force in the past and that plans were in place to address this problem, the Committee remained convinced that the issue of land ownership and distribution was crucial to addressing the economic, social and cultural grievances of a substantial segment of the population.

- 76. In the cases both of Nigeria and Guatemala, the reports addressed the situation under Governments which have since been replaced. Hopefully, the new Governments will take serious and meaningful steps to redress the situation. They should as soon as possible adopt the strategy plans recommended by the Committee and work closely with the relevant international agencies to ensure the elimination of hunger caused by past mismanagement, and ensure the enjoyment for all of adequate food.
- 77. The relationship between sanctions and the right to food has received considerable attention, particularly in the case of Iraq, and was addressed by the Committee on Economic, Social and Cultural Rights in its general comment 8, (E/C.12/1997/8) adopted at its seventeenth session. The Committee made an important point by emphasizing that in considering sanctions, it is essential to distinguish between the basic objective of applying political and economic pressure upon the governing élite of the country to persuade them to conform to international law, and the collateral infliction of suffering upon the most vulnerable groups within the targeted country. The obligations under the Covenant remain applicable even when it has been decided for purposes of international peace and security to impose sanctions.
- 78. Obligations exist both for the target State and for those participating in the imposition and application of sanctions. The State which is the target of sanctions remains obliged under the Covenant to take steps "to the maximum of its available resources" to provide the greatest possible protection for the economic, social and cultural rights of each individual living within its jurisdiction and to ensure that there is no discrimination in relation to the enjoyment of these rights. The target State must take all possible measures, including negotiations with other States and the international community, to reduce to a minimum the negative impact upon the rights of vulnerable groups within the society.
- 79. But, as pointed out by the Committee, obligations apply also to actors responsible for the imposition, maintenance or implementation of the sanctions, whether it be the international community, an international or regional organization, or a State or group of States. Economic and social rights must be taken fully into account when designing an appropriate sanctions regime. Effective monitoring should also be undertaken as long as the sanctions are in force, to protect the economic, social and cultural rights of the affected population. Those who are imposing the sanction must "take steps, individually and through international assistance and

cooperation, especially economic and technical" in order to respond to any disproportionate suffering experienced by vulnerable groups within the targeted country.

Committee on the Rights of the Child

- 80. The monitoring carried out by the Committee on the Rights of the Child (CRC) is also relevant here. States have under article 27 recognized the right of every child to an adequate standard of living, and under article 24, States have committed themselves to combat malnutrition through, inter alia, the provision of adequate nutritious foods and clean drinking water. The Committee in its examination of States parties' reports frequently addresses the situation of food and nutrition for children. It is also concerned with the pre-natal and post-natal health of mothers, and underlines the significance of breastfeeding as a major requirement in avoiding malnutrition and disease in children. Two examples show the direction of its concerns.
- 81. In the concluding observations of the Committee on the Rights of the Child relating to the report by Belize (CRC/C/15/Add.99 of 10 May 1999), the Committee noted with concern the health situation of children in the State party and expressed its concern at the high child and infant mortality rates, poor breastfeeding practices, the high rate of malnutrition, the increasing incidence of stunting and the limited access to safe drinking water, especially in rural communities. The Committee encouraged the State party to develop comprehensive policies and programmes to reduce the incidence of child and infant mortality, to promote and improve breastfeeding practices, to prevent and combat malnutrition, especially in vulnerable and disadvantaged groups of children, and to consider requesting technical assistance for the Integrated Management of Childhood Illnesses and other measures for child health improvement from, inter alia, UNICEF and WHO.
- 82. Thailand is a country in which considerable progress has been recorded, but problems still remain. In the concluding observations relating to the report of Thailand (CRC/C/15/Add.97 of 26 October 1998), the Committee noted the efforts of the State party to reduce the child and infant mortality rates, but expressed its concern at the persistence of poor breastfeeding practices and the high rate of malnutrition. The Committee encouraged the State party to develop comprehensive policies and programmes to promote and improve breastfeeding practices, to prevent and combat malnutrition, especially among vulnerable and disadvantaged groups of children, and consider seeking technical assistance for the Integrated Management of Childhood Illnesses and other measures for child health improvement from, inter alia, UNICEF and WHO.

Committee on the Elimination of Discrimination against Women

83. The Committee on the Elimination of Discrimination against Women (CEDAW) also has important tasks to perform in regard to the rights to food and nutrition. It has been shown that cultural traditions in some parts of the world cause women, particularly the girl child and women of reproductive age, to have less access to adequate food than the boys and men. This has serious consequences both for the women themselves and for the children born to them,

since inadequate nutrition in the pre-natal and post-natal periods will cause malnutrition in the child, whether boy or girl, the consequences of which can last an entire lifetime.

84. The Committee should therefore monitor the implementation of article 5, of the Convention by which States have undertaken to modify social and cultural patterns of conduct to eliminate practices based, inter alia, on stereotyped roles for men and women. Traditionally based sex roles lead in some countries to serious inequalities in the access to adequate food with far-reaching consequences for malnutrition, as pointed out in chapter II of this report. Systematic information and education at school and through other channels should demonstrate the risk and dangers of practices and roles which cause malnutrition in women of reproductive age and particularly during and after pregnancy. Article 12, paragraph 2 requires that adequate nutrition during pregnancy and lactation be ensured. The Committee should, in reviewing States parties' reports, require information specifically on the implementation of these provisions and draw on the material collected by FIVIMS. On that basis it should call on States to report on measures taken to ensure gender equality in access to food and that the special needs of women during pregnancy and lactation are safeguarded.

VII. THE ROLE OF THE COMMISSION ON HUMAN RIGHTS AND ITS SPECIAL RAPPORTEURS

- 85. The Commission on Human Rights in its most recent resolution on the right to food, resolution 1999/24, reaffirmed that hunger constitutes an outrage and a violation of human dignity, and called for the adoption of urgent measures at the national, regional and international level. It has not, however, systematically addressed the right to food or other aspects of the right of everyone to an adequate standard of living, except in the context of the right to development (see further chapter IX below). Nor so far has a thematic rapporteur been appointed to address this issue. Economic and social rights had generally received limited systematic attention by the Commission until in 1998 it appointed a special rapporteur on the right to education, which was a very welcome step. Hopefully, this is the beginning of more consistent attention to the implementation of economic and social rights.
- 86. More encouragingly, some of the special rapporteurs on country situations have in recent years started to give attention not only to civil and political rights but also to economic and social rights. For purposes of illustration, four of the reports presented at the fifty-fifth session of the Commission in 1999 are briefly examined here: those on Afghanistan, Iraq, Myanmar and Nigeria.
- 87. In his report on the situation of human rights in Afghanistan (E/CN.4/1999/40), the Special Rapporteur, Mr. Kamal Hossain, pointed to the widespread loss of human life, destruction of social and economic infrastructure, environmental degradation, food insecurity and malnutrition. Food consumption per capita was still lower than before the war. Chronic malnutrition existed in most parts of Afghanistan, and the infant and maternal mortality rates were among the highest in the world. In most of the country, women suffered deprivation due to widespread poverty, low literacy levels, limited opportunities to participate in public life, limited availability of

health care facilities and restrictions on the employment of women in urban areas. The country had acquired the dubious distinction of being the world's largest producer of narcotic substances.

- In his report on the situation of human rights in Myanmar (E/CN.4/1999/35), the Special Rapporteur, Mr. Rajsoomer Lallah, noted that prisoners were said to be denied adequate food (in amount and quality) and health care. Violence against civilians was a fundamental component of the overall military strategy of the Myanmar army and designed not only to secure resources from the local population, in particular food, combatants and workers, and to weaken the resource base of insurgent groups and their capacity to govern by systematically destroying most of the villages, but also, as the local economy is largely based on agriculture, forcibly to displace the rural population so as to disrupt agricultural production. Testimonies indicated that the armed forces often resorted to searches, destruction and burning of houses and confiscation of property and food to force the people to move, temporarily or permanently. Because of the massive displacement of the population, farming had apparently been drastically reduced. This has led to food shortages among the rural population generally and has provoked voluntary displacement to other areas in search of assistance and security. Apart from the lack of security, the lack of food was clearly the main problem facing families living in the jungle. Families supplemented their rice porridge by eating wild roots, leaves and fruit. Many families interviewed said that they could not cultivate their crop safely and that when they were discovered their fields and paddies were burned. As a result, without any source of income and without enough food to survive, they tried to find their way to Thailand.
- 89. In his report on the situation of human rights in Nigeria (E/CN.4/1999/36), the Special Rapporteur, Mr. Soli Jehangir Sorabjee, dealt at some length with economic, social and cultural rights. His 1999 report came after the change in Government from a period of authoritarian military rule with drastically negative consequences. Mr. Sorabjee noted that the new Government was faced with a daunting challenge, in response to which it had put in place a policy which reflected greater concern for fundamental economic, social and cultural rights. At present the majority of the population did not have access to health facilities, education, food, shelter, or basic amenities such as safe water. The majority of Nigerians were living well below decent social thresholds. The underfunding of the health sector, a situation which had continued for over a decade, meant a steady deterioration in health services. The infant mortality rate in Nigeria had reached 84 per 1,000 and the maternal mortality rate was 800 per 100,000.
- 90. The situation in Iraq has given grounds for particular concern since the country for many years has been the subject of economic sanctions, which have contributed to serious malnutrition in the country. It is a matter of some controversy, however, where the responsibility for this state of affairs lies. In his report to the Commission at its fifty-fourth session in 1998 on the situation of human rights in Iraq (E/CN.4/1998/67), the Special Rapporteur, Mr. Max van der Stoel, referred to the general comment made by the Committee on Economic and Social Rights in December 1997 on sanctions. He pointed out that the Government of Iraq remained responsible for implementation of its obligations relating to the rights to food and health care to the maximum of

its available resources. He further argued that if the Government of Iraq had not waited five years to decide to accept the "oil-for-food" agreement proposed as early as 1991 in Security Council resolutions 706 (1991) and 712 (1991) to meet the humanitarian needs of the population, millions of innocent people would have avoided serious and prolonged suffering. He also referred to reports stating that the effect of the embargo was harsher on members of ethnic and religious minorities, and that there existed discrimination in the allocation by the authorities of the limited resources available between rural and urban areas, and against the southern region with respect to the Marsh people. He therefore concluded that while he was aware of the slowness of United Nations procedures for processing purchase contracts, the Government of Iraq also had the responsibility to assist the approval process.

- 91. In the view of the present author, the conclusion to be drawn from the example of Iraq is that both the Government and the international community have the responsibility to ensure that sanctions do not cause serious violations of human rights, in particular the rights to food and to be free from hunger, and that the responsibility of the international community remains even when the target Government does not cooperate or when it discriminates between groups in their access to food; those who have initiated the sanctions remain responsible for taking corrective measures to ensure access to food.
- 92. While the four country reports mentioned above have given attention to the right to food and nutrition, in many other cases this is not done even where it could have been expected. The Commission should explicitly include in the mandates for the special rapporteurs, wherever appropriate, the task to examine also the situation with regard to this right.

VIII. THE EVOLVING ROLE OF UNITED NATIONS BODIES, SPECIALIZED AGENCIES AND NON-GOVERNMENTAL ORGANIZATIONS

One of the main purposes of the United Nations as listed in Article 55 of the Charter is to promote higher standards of living. Priority should be given to ensure that those presently vulnerable reach an adequate standard of living or at least that they are free from hunger. Another major purpose of the United Nations is to promote universal respect for and observance of human rights. These two tasks are closely intertwined; indeed, social development is increasingly seen and understood as the realization of all human rights for all. The reasoning underlying the human rights approach in this field can be exemplified by a statement by Søren Jessen-Petersen, Assistant United Nations High Commissioner for Refugees. In the area of refugee protection and assistance, as in other fields, the added value of the human rights approach is that it ensures that humanitarian action is based on the rights of the beneficiaries and is not simply a gratuitous act of charity. The refugee ceases to be a mere "recipient" and becomes a "claimant", with a voice and the power to participate in endeavours to meet their basic needs. The human rights approach underlines the legal obligations of States to meet the basic needs of the most vulnerable individuals (including refugees). It leads to a recognition that the role of humanitarian agencies such as UNHCR is to provide support to States in fulfilling their responsibilities, rather than being a substitute for State action (or inaction). 18/ A similar philosophy underlies

the Declaration on the Right to Development: the individual is the central subject of development and should be the active participant and beneficiary of the right to development (art. 2).

- 94. In recent years, an increasing number of United Nations agencies and other bodies have recognized the necessity of using a human rights framework in their work. This has gone hand in hand with an increasing attention to economic and social rights as being of equal value to civil and political rights. The present chapter looks at the evolving conception by United Nations bodies and specialized agencies of their role in promoting the human right to food and nutrition.
- 95. FAO is the lead agency in matters relating to food. It was the host of the World Food Conference in 1974, co-organizer with WHO of the International Conference on Nutrition in 1992, and organizer of the World Food Summit in 1996, playing a major role in the follow-up of the commitments made during the last. The preamble to the FAO Constitution sets "ensuring humanity's freedom from hunger" as one of the basic purposes of the organization. The rights related to food are therefore of special concern to FAO. 19/ During the celebration of the fiftieth anniversary of the Universal Declaration of Human Rights in 1998, FAO produced an excellent booklet on the right to food, entitled "The Right to Food in Theory and Practice", which provided information about the activities of FAO itself and other food-related organizations. Among the many other functions FAO can serve in the future is to make use of its considerable expertise and accumulated knowledge concerning legislation in the field of food and agriculture. The organization is in an excellent position to assist States in the drafting of the national framework legislation requested by the Committee on Economic, Social and Cultural Rights in its general comment 12 (para. 29).
- 96. One of the important bodies under FAO is the Committee on Food Security (CFS), whose concerns include the elimination of food insecurity wherever it may occur. The CFS played a major role in the preparation of the Rome Declaration and World Summit Plan of Action and monitors the implementation by States of their commitments made at the Summit. The CFS at its twenty-fifth session in 1999 commended the collaboration between FAO and the Committee on Economic, Social and Cultural Rights and welcomed the proposal that the cooperation between these institutions be strengthened on a continuing basis.
- 97. One of the most pressing tasks is to identify the hungry and the particular causes of their hunger. The Food Insecurity and Vulnerability Information and Mapping System (FIVIMS) now being developed by FAO in collaboration with other agencies is intended to find out exactly that, in close collaboration with national FIVIMS. Precise identification, using the human rights framework, of the food-insecure or vulnerable groups who they are, where they are located and the particular causes underlying their vulnerability will vastly improve the possibility of developing precise and appropriate responses to those particular situations.
- 98. The International Fund for Agricultural Development (IFAD) assists developing countries in combating rural poverty and improving food production and nutritional levels through investment in poverty alleviation. Part of its activity has been to strengthen the position of rural women, who traditionally

often have faced discrimination in the economic field. Article 14 of the Convention on the Elimination of All Forms of Discrimination against Women requires States to ensure that rural women have the right to agricultural credit and loans, marketing facilities, appropriate technology and equal treatment in land and agrarian reform and resettlement schemes. IFAD projects have sought to support women's productive activities by providing small credit together with training and access to improved and appropriate technology. A stronger reliance on the provisions of the Convention and other human rights instruments would further strengthen the role of IFAD in these endeavours.

- 99. The World Food Programme (WFP) is the food aid agency of the United Nations with a devotion to the protection of freedom from hunger. A human rights approach is important for its work. The WFP increasingly faces conflict situations where hunger has become a weapon of war and where civilians increasingly are subjected to starvation as part of armed conflict. War-affected populations are denied access to food along with other types of humanitarian assistance; WFP seeks to help but often under very difficult circumstances. The prevention of access to humanitarian food aid in internal or international conflicts is a violation of the right to food.
- 100. Armed conflict is not, however, the main cause of hunger. Hundreds of millions of people suffer from the "silent emergency" of chronic hunger, which is a violation or neglect of human rights in itself, and which also causes difficulties for the enjoyment of other human rights. Those affected cannot participate in the development process. Hunger keeps them weak, beset by disease and often lethargic, and when families use most of their time or income to obtain food, they have little possibility for educational or economic activities which could lift them out of their poverty. For these people to have a chance to become active and creative partners in development, timely food aid can be of great help. The WFP approach is to use food aid as a pre-investment to enable marginalized people to take up development opportunities. The food aid gives them temporary food security so that they can get started in the process of helping themselves. It is short-term help that can lead to long-term progress. 20/
- 101. Refugees and displaced persons who have been violently uprooted by conflict, persecution and human rights abuses are particularly at risk of being denied access to adequate food. UNHCR and WFP have signed a memorandum of understanding which covers cooperation in the provision of food aid to refugees, returnees and, in specific situations, internally displaced persons.
- 102. UNICEF has taken an explicit political decision to make the realization of the rights contained in the Convention on the Rights of the Child and the Convention on the Elimination of All Forms of Discrimination against Women the heart of its mandate. The determinants of good nutrition for children food, care and health are covered in the former, particularly article 24. The latter contains similar provisions with respect to the health and nutrition of women. UNICEF uses these two human rights instruments in its struggle against hunger and malnutrition by emphasizing the responsibility of Governments and other actors arising from these conventions, and by a reorientation of the understanding of the nutrition problem. A rights-based approach does treat malnutrition as a purely technical problem, but as a set of violations or as neglect.

- 103. UNICEF holds that rights-based programming and policy development carries a commitment to real participation of those affected. Stephen Lewis, the Deputy Executive Director of UNICEF, argues that when malnutrition is understood as a violation of human rights, it has profound implications for the way the United Nations and government partners do business. It should change the approach to adults (mostly women) and children who are sufferers or potential sufferers of malnutrition. It should improve the effectiveness and sustainability of actions undertaken. Respect, protection, facilitation and fulfilment of nutrition rights are also linked to the evolving concept of good governance at the global, national and local levels. 21/
- 104. WHO is now taking a stronger interest in a human rights approach to food and nutrition as an important element in the right to food. The new Director-General, Gro Harlem Brundtland, has expressed strong commitments in this direction. 22/ "A malnourished child is not only more vulnerable to disease. Cognitive development will be in peril, especially during the first three years of life. Stunted physical growth may therefore be closely linked to reduced mental development. With a distorted intake of nutrients - too little, too much, or unsafe - or with too great a loss of nutrients for example through diarrhoea or errors of metabolism - we easily fall prey to ill-health and disease ... The proper mix of nutrients under clean and safe conditions must be available to all." She has further noted that "human rights begin at home", which means that the agencies, including WHO, should create an internal working environment to take a human rights approach. Statements made during the recent visits by the Director-General to the food agencies in Rome indicate that WHO is also prepared to join in an intersectoral and inter-agency approach to basic rights. The right to food and its relationship to nutritional health appears a good place to start.
- 105. According to James Lovelace of the World Bank, 23/ the Bank recognizes that the human rights approach to nutrition is an important new narrative of the international development discourse. The Bank is only beginning to explore the implications of the human rights framework for its work. Support for governance reform and support for equitable economic growth are but two aspects of the Bank's comprehensive approach which will strengthen the human rights culture. The intention is to remain focused on action to combat malnutrition, working with countries and partner agencies to significantly increase the global resources allocated to nutrition. The Bank will take action to increase the awareness of country officials about nutrition problems. The Bank will incorporate nutrition into agriculture and rural development projects and food policy reform and ensure that nutrition is part of the design of health projects.
- 106. As pointed out in my 1998 progress report (para. 38), the policy document by UNDP finalized in 1997 and published in January 1998 entitled "Integrating human rights with sustainable human development" has been a major milestone in the ongoing shift towards a human rights approach to development.
- 107. For the technical and strategic roles that the United Nations system as a whole can play in the future work to advance the right to adequate food in the pursuance of nutrition rights, the Sub-Committee on Nutrition of the Administrative Committee on Coordination (ACC/SCN) needs special attention as the focal point for harmonizing policies and activities in nutrition of the

United Nations system. $\underline{24}/$ Besides compiling and disseminating information, SCN promotes coordinated inter-agency activities aimed at reducing malnutrition. The SCN is responsible for overseeing the direction, scale, coherence and impact of the United Nations response to the nutritional problems of the world. The recent Commission on Nutrition Challenges for the 21st Century, whose report is referred to in chapter II of this report, was an initiative of the SCN.

- 108. Given its mandate and scope, the SCN is an important forum in which to channel information and initiatives regarding the human rights approach to the alleviation of hunger and poverty, thereby promoting food and nutrition security. The SCN has a working group on Nutrition, Ethics and Human Rights where, through its successive meetings and dialogue since 1994, member agencies and observers have discussed the potential of such an approach. The value of a human rights approach to food and nutrition policies and programmes was recognized in particular through the successful symposium on this topic hosted by the High Commissioner for Human Rights in Geneva in April 1999, described in chapter III above. Provided appropriate professional and institutional working linkages can be established between OHCHR and SCN, a system-wide assault on violations and neglect of food and nutrition rights the world over can be set in motion, involving many agencies separately and collectively. It would encompass technical standard-setting, development of benchmarks and indicators, and food and nutrition policy guidance - now founded on a human rights platform. It could enhance the quality of reporting to and monitoring by the Committee on Economic, Social and Cultural Rights by advancing the knowledge of the critical factors influencing food and nutrition.
- 109. Another channel for pursuing the right to adequate food should be the ACC Network on Rural Development and Food Security established in April 1997 as the mechanism for inter-agency follow-up to the World Food Summit, jointly managed by FAO and IFAD in close cooperation with WFP. The Network currently includes 20 interested United Nations organizations which participate in and support country-focused and demand-driven thematic groups composed of the concerned partners, including United Nations agencies, national institutions, bilateral donors and civil society organizations, setting agendas and implementing activities that meet national needs and priorities.
- 110. Research institutions can also have important roles to play in the advancement of the right to food. The International Food Policy Research Institute (IFPRI), headquartered in Washington, requires special mention. IFPRI focuses on research to help foster sustainable economic growth and combat poverty through better government policies. Aiming at reducing hunger and malnutrition, IFPRI's interest extends to social, economic and institutional forces that drive the food sector and bear on the development process. A major recent IFPRI initiative is the 2020 Vision for Food, Agriculture, and the Environment to "identify solutions for meeting future world food needs while reducing poverty and protecting the environment". IFPRI is in a special position to conduct and promote research and policy to promote the right to adequate food from a human rights perspective. A first IFPRI policy brief on the right to food was issued in 1995, and a new brief will be forthcoming in 1999 under the 2020 initiative.

- 111. The role of the civil society and the non-governmental sector will be crucial in advancing freedom from hunger and the right to adequate food. Non-governmental human rights organizations have for a long time concentrated mainly on civil and political rights, whereas non-governmental development organizations have given little attention to human rights. In recent years, this has changed. An increasing number of NGOs and social movements have started to use the human rights framework and this is likely to increase with the growing precision of the content of economic and social rights. As to the right to food, Michael Windfuhr, representing one of these organizations, has pointed out that "... many NGOs are well aware that promotion of the right to food requires a clear definition of the basic concept of economic, social and cultural rights and a continuous 'unlearning' of the reductionist human rights concept, which is focused entirely on civil and political rights, as well as the reductionist concepts of food security that focus only or mostly on agricultural productivity and harvest yield". 25/
- 112. NGOs took a great interest in the World Food Summit. More than 1,000 organizations from more than 80 countries attended the parallel NGO Forum during the Summit. One of their demands was the preparation of a code of conduct on the right to adequate food. In objective 7.4 of the Summit Plan of Action, the World Food Summit opened up the possibility of preparing "voluntary guidelines" for promoting the right to food and food security. FoodFirst Information and Action Network (FIAN), the World Alliance for Nutrition and Human Rights and the Jacques Maritain Institute took the initiative and have jointly drafted such a code in consultation with a large number of other NGOs worldwide. It sets out the normative content of the human right to adequate food, the corresponding obligations including State obligations at the national level and at the international level, the responsibilities of international organizations, and regulation of economic enterprises and other actors. It deals with the national framework for monitoring and recourse procedures, and with international reporting and support mechanisms. The draft code also sets out responsibilities of the actors of civil society. The NGOs involved intend to place the proposed code of conduct on the agendas of the Commission on Human Rights and the FAO Committee on Food Security. $\underline{26}$ / Another NGO initiative, the Global Forum for Sustainable Food and Nutritional Security, established as a post-World Food Summit network to ensure continued debate and mutual support especially among NGOs in the South, can be a critical factor in mobilizing more NGOs for the right to food the world over.
- 113. A whole set of special problems arise for agencies and organizations involved in humanitarian action during armed conflict. These difficult issues are not addressed by the present study. It would be highly desirable that a separate study or report be prepared on the implementation in times of armed conflict of economic and social rights, including the right to food. Suffice it here to say that established humanitarian actors such as the International Committee of the Red Cross and the Federation of Red Cross and Red Crescent Societies have been joined both by several intergovernmental organizations and by a multitude of NGOs. Each of them has their own specific mandates and their particular ethical and moral perspectives. Coordination and accountability have become matters of great difficulty.

IX. GLOBALIZATION AND THE RIGHT TO FOOD

- 114. The main features of globalization have been an increasing reliance on a deregulated global market, maximally free trade, uncontrolled flow of investment, and the paramount role of the international financial market, with the IMF and the World Bank moving to the centre stage of world power. In a wider perspective, the content of globalization is also affected by developments in science and technology, communications and, in particular, information-processing, which has substantially changed the structure of the global system. Positive aspects of globalization therefore include advances in communication and information and therefore much greater openness.
- 115. We have, however, seen too much of the downside of globalization. Imposed requirements of structural adjustment and the burden of debt are both part and parcel of the globalization process. The policies of structural adjustment pursued in the 1980s probably contributed in many places to intensified tension between the different ethnic, racial or social groups in society, and may have had enormous cost in terms of lost development opportunities and may have hampered many groups in their enjoyment of the right to food. In recent years, the policies of structural adjustment have been improved but continue to carry risks for the economic and social rights of the most vulnerable part of the population.
- 116. The Committee on Economic, Social and Cultural Rights on 11 May 1998 adopted a statement on globalization (E/1999/22, paras. 515-516) in which the Committee, reflecting on the downside of globalization, argued that its negative risks can be guarded against, or compensated for, if appropriate policies are put in place, but that insufficient efforts are being made to devise new or complementary approaches which could enhance the compatibility of those trends and policies with full respect for economic, social and cultural rights. As noted by the Committee, Governments and intergovernmental and international organizations have a strong and continuous responsibility to take whatever measures they can to act in ways compatible with their human rights obligations. The areas of trade, finance and investment are not exempted from such responsibilities
- 117. Globalization should also mean global responsibility, including global responsibility to assist in the creation of conditions for the full enjoyment of human rights. It is in this context that the right to development has its main significance, with its dual emphasis on domestic implementation of all human rights and international cooperation to make that possible. While on the one hand every Government should remove obstacles blocking development within its State, such as failure to observe civil, political, economic, social and cultural rights, effective international cooperation is essential in providing developing countries with appropriate means and facilities to foster their comprehensive development, in accordance with the Declaration on the Right to Development.
- 118. Globalization of the market should be regulated through application of environmental standards and universal human rights, through determined action both by the United Nations and by States. While WTO, the IMF and, to some extent, the World Bank are institutional agents of globalization, the United Nations, in particular its human rights bodies but also other agencies

such as the UNDP, are now agents of the universalization of rights. A constructive and corrective relationship between the forces of the market and the powers of the State requires good governance, which means that it shall be so structured that it optimally implements human rights. It requires rule of law, transparency, responsiveness and accountability at the national level, and it also requires rule of law, transparency and, in particular, responsiveness by the international community.

- 119. It is within this framework that we must consider also the prospects for the right to adequate food: the potential for and constraints to its universalization. Among the critical evolving issues are the effects of liberalized trade and the negotiations over trade barriers including standards set to protect the health of consumers. Another set of issues are those related to the revolution in biotechnology and the opportunities for modifying and transferring plant or animal gene material between organisms both within and across species frontiers for more effective food production, whether facilitated by public or private-sector research.
- 120. Derived from these innovations is the particular claim for intellectual property rights (IPRs) in connection with engineered new forms of live material. In the case of new varieties of plants this has come mainly in the form of "plant breeders' rights" 27/ though some countries also allow industrial patents for biological materials. The human rights concern with respect to IPR has found its place in the realm of cultural rights. The merit of IPR from a human rights perspective becomes less obvious when they are used by powerful economic actors as a tool to control the economically weaker producers of food for human consumption (or of other agricultural products that indirectly affect the distribution of the means of production of food). There has been growing concern that intellectual property rights as presently defined may seriously threaten the pattern of and control over our future food supply. It may therefore be time that IPR be studied by the human rights community in order to outline potential conflicts of interests that may seriously jeopardize the right to adequate food.
- 121. Another set of rights and interest groups to claim them outside the international human rights system are farmers' rights, currently being debated in international forums. 28/ The concept of farmers' rights is based on the notion that farmers, both historically and at present, have made great contributions to the conservation and development of agricultural biodiversity. Whereas commercial interests have recourse to rights such as IPRs, farmers typically have no formal rights and no protection associated with their activities. Farmers' rights should be given attention by the human rights community and promoted in the continued promotion of the right to food, since our future food supply and its sustainability may depend on such rights being established on a firm footing.
- 122. Finally, "consumer rights" is still another set of rights outside the international human rights system and claimed by a particular interest group. Because all human beings are also consumers, it may be useful to explore how international consumers' rights and duties, as established by a large international consumers' organization 29/ and backed up by United Nations-based guidelines for consumer protection, can help in the realization of the right to adequate food as a human right. Consumer groups

may benefit from using human rights norms, particularly those regarding the right to adequate food. Consumer rights can also serve as references in the interpretation of treaty-based provisions relevant to the right to food and nutrition.

- 123. The process of globalization as it affects food has many and complex features and we may be just at the beginning of understanding its long-term effects. In a situation where there is no agreement, even among technical experts, on whether genetically modified organisms (GMO), now slowly entering the food manufacturing arena, are dangerous to health; whether they should be clearly indicated on the label; whether they are being adequately monitored by quality-control systems; and whether proof should be given that food containing GMO is safe before it is marketed. The notion of the global supermarket born in the late 1970s takes on truly new dimensions. At the time it described the growing uniformity of dietary patterns and the threats to small farmers and food-processing enterprises, and a trend towards monocultures and large processing conglomerates. As if this were not problematic enough, the added safety dimensions and the potential implications for health of the globalizing supermarket warrant closer attention to be paid to the means used to realize the universal right to adequate food.
- 124. In discussing possible future trade-off between health (and other so-called non-trade) values as "trade barriers" and guaranteeing access to safe food as an integral part of the right to adequate food, paragraph 36 of general comment 12 on the right to adequate food should be borne in mind. This was discussed in chapter IV but is worthy of completing this chapter: "States parties should, in international agreements whenever relevant, ensure that the right to adequate food is given due attention and consider the development of further international legal instruments to that end." In the forthcoming new round of trade negotiations within WTO, concerned States and NGOs may do well to keep this statement in mind.
- 125. I have pointed to some of the issues which need to be addressed in future work on the right to adequate food. The answers are not clear at this stage, however, in line with the statement on globalization in 1998 by the Committee, it is incumbent on the international community to guard against the negative risks. This must be done concurrently with pursuing approaches which can improve the access to food for those who are most vulnerable.

X. CONCLUSIONS AND RECOMMENDATIONS

Conclusions

- 126. Three major points emerge from this update of the study on the right to adequate food:
 - Firstly, that since 1987 there has been a much better and more widely shared understanding of the impact of hunger and malnutrition, and a more general awareness that their consequences are far worse than was previously recognized;

- Secondly, that there is a much broader recognition of the right to be free from hunger and to adequate food as human rights. The adoption by the World Food Summit in 1996 of its Plan of Action was a major watershed in attitudes towards this issue. The adoption by the Committee on Economic, Social and Cultural Rights in May 1999 of general comment 12 has contributed significantly to the clarification of the content of the right and the corresponding State obligations. The consultations conducted by the High Commissioner for Human Rights and the symposium organized by the ACC/SCN have deepened and broadened that recognition;
- Thirdly, that international institutions now broadly endorse the human rights approach to food and nutrition issues and have pledged themselves to act, separately and collectively, to contribute to the realization of these rights. The information given in chapter VIII about the commitments by UNICEF, FAO, WHO, WFP, IFAD and UNHCR, and the involvement of bodies such as the ACC/SCN, show that there is a new and potentially powerful momentum to act in a concerted way to eliminate the scourge of hunger from humanity.
- 127. The two main questions which remain are these: Will States, in light of the urgency of the matter, take all possible steps within their available resources to bring hunger to an end and to move towards full realization of the right to food? Secondly: Are other States, and thereby the international community, prepared to shoulder what is at least their moral obligation to respect and protect food and nutrition security in other countries and contribute as necessary, individually and collectively, to facilitating and fulfilling such security in countries in need?
- 128. The next few years will show whether States are prepared to move from rhetoric to realization of these rights which, if done in a concerted way, is fully within the power of the international community. Without energetic implementation both at the national and the international level, however, the insight gained will not lead to the aims set; at a minimum this includes the reduction of the number of undernourished people to half their 1996 level by 2015.

Recommendations

- 129. Many of the recommendations which follow have their origins in the consultations convened by the High Commissioner and the ACC/SCN symposium held on 12 and 13 April 1999. $\underline{30}/$
- 130. Recommendations concerning national implementation by States:
- (a) States should elaborate a strategy for the realization for everyone within their territory of freedom from hunger and the right to adequate food. The strategy should include the elaboration of framework legislation as proposed in general comment 12, setting out its purpose; the targets or goals to be achieved and the time-frame to be set for the achievement of those targets; the means by which the purpose could be achieved described in broad terms, in particular the intended collaboration with

civil society and the private sector and with international organizations; institutional responsibility for the process; and the national mechanisms for its monitoring;

- (b) As part of the strategy, States should develop or expand national indicators on food insecurity in line with the FIVIMS initiative coordinated by FAO (see chapter VIII). This system is intended to absorb data from various bases within other United Nations agencies including the comprehensive nutrition-related data set compiled by UNICEF in some regions;
- (c) Other essential parts of the strategy are (i) to develop and maintain mechanisms to monitor progress towards the realization of the right to adequate food for all, and it should for this purpose establish appropriate mechanisms for handling the technical food and nutrition dimensions of domestic and international reporting; (ii) to identify the factors and difficulties affecting the degree of implementation of their obligations; and (iii) to facilitate the adoption of corrective legislative and administrative measures. This is important also where the country concerned has demonstrable resource constraints and where international assistance is therefore required to supplement exhausted domestic resources;
- (d) As recommended by the ACC/SCN Commission on Nutrition Challenges for the 21st Century, national nutrition councils should be established in all countries; as part of their agenda they should expand the constituency of analysts and practitioners who can help advance the nutrition message and mission through the perspective of human rights and strengthen national nutrition monitoring and surveillance systems, improve methods of analysis and expand this analysis to include issues of accountability, contribute to improved national reporting on the nutrition situation, promote national endorsement of the draft international code of conduct on the right to adequate food, and promote dialogue between the nutrition and human rights communities so as to strengthen both communities and help provide the knowledge-based and operational skills needed to advance the human right to food and nutrition.
- 131. Recommendations to States regarding international cooperation and prevention of deprivation:
- (a) States should recognize the essential role of international cooperation and comply with their commitment to take joint and separate action to achieve the full realization of the right to adequate food;
- (b) States should respect the enjoyment of the right to food in other countries; among other things this means that they should at all times refrain from using food as a weapon. States participating in international sanctions should ensure that they do not lead to the deprivation of necessary food for the population;
- (c) States should protect the right to food in other countries which, <u>inter alia</u>, requires that they should monitor the impact of their external aid and trade, as well as the impact of transnational corporations under their jurisdiction, on the enjoyment of the right to food in other countries and take corrective measures where a negative impact can be shown;

- (d) States should facilitate access to food in countries other than their own, and provide the necessary aid when required. This does not necessarily or only mean food aid, but various categories of aid which enable the local population concerned to develop sustainable and effective entitlements to food either by its own production or by income generation. Donor States should increase their awareness of the importance of the effective enjoyment of the right to food, and make it a major concern in their development assistance;
- (e) Creditor countries should adopt substantial debt relief measures in ways which facilitate the realization of the right to food and other economic and social rights for vulnerable groups in developing countries;
- (f) States parties to international agreements, whenever relevant, should ensure that the right to adequate food is given due attention and consider the development of further international legal instruments to that end.
- 132. Recommendations to the High Commissioner for Human Rights:
- (a) The High Commissioner for Human Rights should continue to take the leadership role in promoting the implementation of the right to food and nutrition as a human right and to develop a comprehensive strategy for this purpose. Elements of such a strategy would include dialogue with States supplemented by advisory services and other forms of assistance. Partners in the elaboration and implementation of the strategy should include both the legal and the political human rights bodies, as well as the specialized agencies and other bodies, including NGOs;
- (b) In order fully to respond to the request by the World Food Summit and taking into account that the general comment has contributed to the clarification of the right to food, the High Commissioner should now organize a broad consultation to draw up a coherent United Nations strategy for the implementation of freedom from hunger and the enjoyment of the right to food;
- (c) The High Commissioner should strengthen her Office's capacity to deal with the substantive issues of the right to food, including the possibility of appointing an external adviser on the right to food in order to cooperate effectively with the Rome-based food agencies and others involved;
- (d) The High Commissioner should promote a coordinated approach to the right to adequate food throughout the United Nations system, making use for this purpose, <u>inter alia</u>, of the United Nations Development Assistance Framework (UNDAF);
- (e) The High Commissioner should take appropriate steps to improve the professional, institutional and operational links between the concern for human rights and the mandates of the specialized agencies;

- (f) Taking into account general comment 10, the High Commissioner should take steps to set in motion a study on the impact of globalization on the food and nutrition situation of vulnerable groups and on remedies which could be adopted in order to reverse negative impact;
- (g) The High Commissioner should take steps to organize a third Expert Consultation following those in 1997 and 1998, this time with a focus on implementation mechanisms at country level, inviting government experts to bring experiences from their countries as a contribution to the operationalization at national level, including the drawing up of a framework law;
- (h) The High Commissioner should also seek to organize a consultation involving, on the one hand, representatives of the treaty bodies, food agencies and humanitarian organizations and, on the other hand, the international financial and trade institutions, to enhance dialogue and encourage consultations on a regular basis and to prevent activities which would cause a further deterioration in the enjoyment of the right to food and other economic, social and cultural rights;
- (i) The High Commissioner should initiate a study or similar activities intended to clarify the joint and separate responsibility of States, in accordance with the Charter of the United Nations, to cooperate in ensuring access to food for all in times of emergency with special attention to the situation during armed conflicts. The study could examine questions such as:
 - (i) What are the gaps in humanitarian, refugee and human rights law relating to the protection of human rights, including the right to adequate food, in the prevailing environment of today's acute crises?
 - (ii) How can the complementarity between these branches of international law be brought to bear more effectively?
 - (iii) How can the principle of State sovereignty be reconciled with such concepts as the right to assistance in complex emergencies and the right or duty to provide assistance?
 - (iv) How can the ethical and political conflicts often arising between the provision of food and other humanitarian assistance be reconciled with the protection of human rights in acute crises?
 - (v) Should the humanitarian imperative of meeting food and other essential necessities prevail under all circumstances in complex emergency situations?
 - (vi) Is there a need for new standard-setting on humanitarian assistance, and what would be its principal features? How would it affect the human right to adequate food?

(vii) How can conflict prevention, resolution, reconciliation and reconstruction be better integrated with the help of international human rights and humanitarian law?

133. Recommendations to treaty bodies:

- (a) The Committee on Economic, Social and Cultural Rights, CEDAW and the Committee on the Rights of the Child should explore ways and means to establish cooperative links among themselves and with the specialized agencies. Working relationships should be institutionalized between these committees and the ACC/SCN with a view to ensuring the flow of information between them;
- (b) The Committee should periodically or when need arises review its guidelines for State reporting. The Committee should draft new guidelines on the right to food, taking into account the content of its own general comment and the new understanding of the nature and consequences of hunger and malnutrition;
- (c) In their review of State parties' reports, treaty bodies should request information about whether persons or groups who are victims of a violation of the right to adequate food have access to effective judicial or other appropriate remedies and whether they are entitled to adequate reparation.
- 134. Recommendations to the Commission on Human Rights:
- (a) Whenever appropriate, the mandates of special rapporteurs on country situations should include investigations on violations of the right to adequate food;
- (b) The Commission should envisage the appointment of a thematic rapporteur on freedom from hunger and the enjoyment of adequate food;
- (c) The Commission should put on its agenda the question of adopting voluntary guidelines on the implementation of freedom from hunger and the right to food, taking into account, <u>inter alia</u>, the draft code of conduct on the human right to adequate food prepared by three non-governmental institutions and now endorsed by more than 800 organizations.
- 135. Recommendations to the specialized agencies and other bodies:
 - (a) Recognition and training:
 - The agencies should individually define their human rights objectives and establish explicitly their human rights goals in relation to their particular mandate. Secondly, they should use or develop appropriate forums to ensure consensus, compatibility and the complementarity of their respective actions. Joint training and workshops would be helpful in ensuring a common approach;

- (ii) Measures should be taken to strengthen the capacity of the staff to understand, articulate and analyse more effectively the economic, political and social context in which they work, which will be of direct use in their understanding of appropriate strategies in the fight against malnutrition;
- (iii) Recognizing the importance of training for organizations adopting a rights-based approach, efforts to develop training materials on human rights related to food and nutrition should be undertaken or supported. Training materials might need to be tailored to incorporate issues of particular concern to the individual food agency;
 - (iv) The agencies, and the technicians and managers working in them, should learn more about human rights as they directly affect the specialized fields of the agency concerned. The skills should also be strengthened of the field staff in working with communities or finding the most appropriate partners to help communities accelerate their own processes of problem assessment and analysis and the formulation of effective actions to combat malnutrition;
 - (v) Conversely, human rights officers within the United Nations system should receive basic training in the major food and nutrition policy issues;

(b) Indicators and assistance:

- (i) Efforts should be made to better develop indicators and benchmarks concerning the achievements and shortcomings in the realization of food and nutrition rights, particularly by contributing to the Food Insecurity and Vulnerability Information and Mapping System (FIVIMS) and other databases needed for special agency purposes;
- (ii) FAO should offer its assistance, upon request, in drafting national framework legislation and in reviewing the sectoral legislation, taking into consideration FAO's considerable expertise and accumulated knowledge concerning legislation in the fields of food and agriculture. Similarly, UNICEF should make available on request its expertise concerning legislation with regard to the right to adequate food for infants and young children through maternal and child protection including legislation to enable breastfeeding, and to the regulation of marketing of breast milk substitutes;
- (iii) The agencies should continue their efforts to promote participatory and intersectoral approaches to resolving malnutrition everywhere, in particular in women and children but also taking into account the changing patterns of nutritional diseases and groups affected by "the double

- burden of disease", as well as the need to involve whole families, including men, in improving household nutrition;
- (iv) In spite of the continuing difficulties and the new challenges posed by HIV/AIDS, the relevant agencies should continue to support, protect and promote practices of breastfeeding and the conditions required for it;

(c) Cooperation

- (i) FAO, WFP and IFAD should continue to explore how to cooperate more effectively on the implementation of the right to food at the national level, with due respect for their individual mandates and building on their respective expertise. These and other agencies might collectively or individually collect, systematize and make available country cases of best practices on the implementation of the right to food;
- (ii) The ACC Network on Food Security and Rural Development coordinated by the Rome-based agencies should be another vehicle besides the ACC/SCN through which to promote food security and reduce rural poverty from a human rights perspective;
- (iii) Agencies should establish and make use of an internal mechanism to ensure that their own policies and programmes do not have a negative effect on the implementation by States of the right to food and nutrition.
- 136. Recommendations to NGOs, professional organizations and academic institutions:
- (a) Non-governmental organizations should devote more attention to economic and social rights, including the right to food and nutrition. A link should be established between those NGOs which have focused their interest on the activities of the development agencies and those which have focused on the activities of the human rights bodies. Generally, there should be more interaction between human rights and development organizations;
- (b) Associations of judges and lawyers, including bar associations, should train their members in addressing and applying or using economic and social rights on the same level as civil and political rights;
- (c) Universities should as a special responsibility organize training courses in the interface between human rights law and food and nutrition policy analysis and planning, especially as it relates to globalization processes, sustainable agriculture and other development, gender issues and the global burden of disease.

<u>Notes</u>

- 1.Most recently in resolution 1999/24.
- 2. <u>Right to Adequate Food as a Human Right</u>, Human Rights Study Series No. 1, United Nations publication (Sales No. E.89.XIV.2), United Nations, New York, 1989.
- 3. This figure is suggested by the Commission on the Nutrition Challenges for the 21st Century of the Administrative Committee on Coordination Subcommittee on Nutrition (ACC/SCN) led by Professor Philip James, which submitted its final report, entitled "Ending Malnutrition by 2020: an Agenda for Change in the Millennium" in March this year (hereafter "ACC/SCN report").
- 4.In a keynote address by Dr. Brundtland to the ACC/SCN symposium entitled "The Substance and Politics of a Human Rights Approach to Food and Nutrition Policies and Programmes", held at the Palais des Nations in Geneva on 12 and 13 April 1999.
- 5. Further details and country examples are found in the ACC/SCN report, $\underline{\text{supra}}$ note 3.
- 6. Several of the arguments brought forward here are inspired by parts of the ACC/SCN report, supra note 3.
- 7. The ACC/SCN report explains the causal linkages in poor societies as "This life cycle of handicap has now been given a remarkable new slant with the discovery that third world adults moving from impoverished rural areas to seek new opportunities in towns and cities find that, with the marked changes in lifestyle and diet, their weight and health does seem to improve. However, as they put on modest amounts of weight, they selectively accumulate fat within their abdomen and develop a virulent form of abdominal obesity with an increased susceptibility to diabetes, high blood pressure and coronary heart disease. This abdominal obesity seems to explain the explosion in diabetes and cardiovascular disease in the very societies which have been or still are struggling to cope with the major problems of malnutrition. Maternal malnutrition seems to reprogramme the foetus so that in later life, with a change to a high-fat, energy-dense diet and with less physical activity, the adult displays the impact of foetal malnutrition. Obesity already affects 250 million people with a tidal wave of diabetes and visceral obesity emerging throughout the developing world. Most heart disease, strokes and cancers already occur in developing countries; these diet-related diseases now overwhelm health-care systems. There is new evidence that this health burden will accelerate by 2020. Many developing countries have over one-fifth of middle-aged adults with incipient or frank diabetes associated with abdominal obesity."
- 8. The burden of HIV/AIDS, especially in Africa south of the Sahara, adds to the problem insofar as households which are hard hit may completely lose their ability to cope, in some families leaving children and the elderly as the major age groups. While this tragic scenario lies outside the question of the right to food, its impact on the realization of this right cannot be ignored.

- 9.Statement to the ACC/SCN symposium, supra note 4.
- 10. The right to an adequate standard of living is found in article 25 of the Universal Declaration of Human Rights, in article 11.1 of the International Covenant on Economic, Social and Cultural Rights, and in article 27 of the Convention on the Rights of the Child.
- 11. The Maastricht Guidelines on Violations of Economic, Social and Cultural Rights is useful in this respect. They were drawn up in January 1997 by a group of experts, including members of treaty bodies, to assist mechanisms that monitor economic, social and cultural rights. The Guidelines are published in Human Rights Quarterly, vol. 20, No. 1, 1998; in the Netherlands Quarterly of Human Rights, vol. 15, No. 2, 1997, and in SIM Special No. 20, published by the Netherlands Institute of Human Rights.
- 12. See further chapter 8 on the FIVIMS Food Insecurity and Vulnerability Information and Mapping System initiated by FAO.
- 13.FAO, "The Right to Food in Theory and Practice", Rome, 1998, p. 32.
- 14. The notion of food security was addressed at some length in my 1987 study. I criticized the prevailing focus on global food security, which erroneously assumed that there might emerge global deficits in food. In its place, based on the the work by Eide et al. on the notion of household food security, an integrated framework for the implementation of the right to food was developed. That approach has been further refined and increasingly accepted.
- 15. "Understanding human rights approaches to food and nutritional security in Brazil", a document prepared by a working group in Brasilia and presented by Denise Costa Coitinho of the Brazilian Ministry of Health to the ACC/SCN symposium, supra note 4.
- 16.A National Framework for the Promotion and Protection of the Rights to Food Security and Nutrition: A Case Study by the South African Human Rights Commission, presented by Tseliso Thipanyane to the ACC/SCN symposium, $\underline{\text{supra}}$ note 4.
- 17. Paragraphs 21-26 of general comment 12.
- 18. "Food as an Integral Part of International Protection" by Søren Jessen-Petersen, Assistant High Commissioner for Refugees, presented at the ACC/SCN symposium, $\underline{\text{supra}}$ note 4.
- 19. Jacques Diouf, Director-General of FAO, in the Foreword to "The Right to Food in Theory and Practice", <u>supra</u> note 13.
- 20.Statement made by Mr. A. Namanga Ngongi, Deputy Executive Director, World Food Programme, at the ACC/SCN symposium, <u>supra</u> note 4.
- 21.Statement at the ACC/SCN symposium, supra note 4.
- 22. Supra note 4.
- 23. Supra note 9.

- 24. The current members are the United Nations, UNICEF, UNDP, the United Nations Environment Programme (UNEP), the United Nations Population Fund (UNFPA), WFP, UNHCR, the United Nations University (UNU), the United Nations Research Institute for Social Development (UNRISD), the Office of the United Nations High Commissioner for Human Rights (OHCHR), ILO, FAO, UNESCO, WHO, the World Bank, IFAD, the International Atomic Energy Agency (IAEA) and the Asian Development Bank. Bilateral donor agencies participate actively in SCN activities and non-governmental organizations are increasingly involved. The secretariat is hosted by WHO in Geneva.
- 25.Michael Windfuhr, NGOs and the right to adequate food, in "The Right to Food in Theory and Practice", $\underline{\text{supra}}$ note 13.
- 26. For further details about the code, see ibid.
- 27. The International Union for the Protection of New Varieties of Plants, known as UPOV, is an intergovernmental organization with headquarters in Geneva. (The acronym UPOV is derived from the French name of the organization, Union internationale pour la protection des obtentions véqétales.) The Convention for the Protection of New Varieties of Plants (the "UPOV Convention") was signed in Paris in 1961 and latest revised in the 1991 Act which entered into force on 24 April 1998. Its purpose is to ensure that the member States of the Union acknowledge the achievements of breeders of new plant varieties, by making available to them an exclusive property right, on the basis of a set of uniform and clearly defined principles. Protected varieties must, in short, be "distinct, uniform, and stable". Such requirements would disqualify most farmer-developed varieties which, for agronomic purposes, are not uniform or stable, but diverse and changing. Under an earlier (1978) Act, the minimum scope of the plant breeder's right requires that the holder's prior authorization is necessary for "the production for purposes of commercial marketing, the offering for sale and the marketing of propagating material of the protected variety". The 1991 Act contains more detailed provisions defining the acts concerning propagating material in relation to which the holder's authorization is required.
- 28. See FAO resolutions 4/89 and 5/89.
- 29.Consumers International, formerly International Organization of Consumer Unions. See $\underline{\text{http://193.128.6.150/consumers/about/rights.html}}$ for details on consumers' rights and duties.
- 30. The present author participated in all of these (as keynote speaker or panelist, and sometimes also as chairman or rapporteur), and has therefore greatly benefited from these meetings.

<u>Annex</u>

GENERAL COMMENT 12 - THE RIGHT TO ADEQUATE FOOD (art. 11)

<u>Introduction and basic premises</u>

- 1. The human right to adequate food is recognized in several instruments under international law. The International Covenant on Economic, Social and Cultural Rights deals more comprehensively than any other instrument with this right. Pursuant to article 11.1 of the Covenant, States parties recognize "the right of everyone to an adequate standard of living for himself and his family, including adequate food, clothing and housing, and to the continuous improvement of living conditions", while pursuant to article 11.2 they recognize that more immediate and urgent steps may be needed to ensure "the fundamental right to freedom from hunger and malnutrition". The human right to adequate food is of crucial importance for the enjoyment of all rights. It applies to everyone; thus the reference in article 11.1 to "himself and his family" does not imply any limitation upon the applicability of this right to individuals or to female-headed households.
- 2. The Committee has accumulated significant information pertaining to the right to adequate food through examination of States parties' reports over the years since 1979. The Committee has noted that while reporting guidelines are available relating to the right to adequate food, only few States parties have provided information sufficient and precise enough to enable the Committee to determine the prevailing situation in the countries concerned with respect to this right and to identify the obstacles to its realization. This General Comment aims to identify some of the principal issues which the Committee considers to be important in relation to the right to adequate food. Its preparation was triggered by the request of Member States during the 1996 World Food Summit, for a better definition of the rights relating to food in article 11 of the Covenant, and by a special request to the Committee to give particular attention to the Summit Plan of Action in monitoring the implementation of the specific measures provided for in article 11 of the Covenant.
- In response to these requests, the Committee reviewed the relevant reports and documentation of the Commission on Human Rights and of the Sub-Commission on Prevention of Discrimination and Protection of Minorities on the right to adequate food as a human right; devoted a day of general discussion to this issue at its seventh session in 1997, taking into consideration the draft international code of conduct on the human right to adequate food prepared by international non-governmental organizations; participated in two expert consultations on the right to adequate food as a human right organized by the Office of the United Nations High Commissioner for Human Rights (OHCHR), in Geneva in December 1997, and in Rome in November 1998 co-hosted by the Food and Agriculture Organization of the United Nations (FAO), and noted their final reports. In April 1999 the Committee participated in a symposium on "The substance and politics of a human rights approach to food and nutrition policies and programmes", organized by the Administrative Committee on Coordination/Sub-Committee on Nutrition of the United Nations at its twenty-sixth session in Geneva and hosted by OHCHR.

- 4. The Committee affirms that the right to adequate food is indivisibly linked to the inherent dignity of the human person and is indispensable for the fulfilment of other human rights enshrined in the International Bill of Human Rights. It is also inseparable from social justice, requiring the adoption of appropriate economic, environmental and social policies, at both the national and international levels, oriented to the eradication of poverty and the fulfilment of all human rights for all.
- 5. Despite the fact that the international community has frequently reaffirmed the importance of full respect for the right to adequate food, a disturbing gap still exists between the standards set in article 11 of the Covenant and the situation prevailing in many parts of the world. More than 840 million people throughout the world, most of them in developing countries, are chronically hungry; millions of people are suffering from famine as the result of natural disasters, the increasing incidence of civil strife and wars in some regions and the use of food as a political weapon. The Committee observes that while the problems of hunger and malnutrition are often particularly acute in developing countries, malnutrition, under-nutrition and other problems which relate to the right to adequate food and the right to freedom from hunger, also exist in some of the most economically developed countries. Fundamentally, the roots of the problem of hunger and malnutrition are not lack of food but lack of access to available food, inter alia because of poverty, by large segments of the world's population.

Normative content of article 11, paragraphs 1 and 2

6. The right to adequate food is realized when every man, woman and child, alone or in community with others, have physical and economic access at all times to adequate food or means for its procurement. The right to adequate food shall therefore not be interpreted in a narrow or restrictive sense which equates it with a minimum package of calories, proteins and other specific nutrients. The right to adequate food will have to be realized progressively. However, States have a core obligation to take the necessary action to mitigate and alleviate hunger as provided for in paragraph 2 of article 11, even in times of natural or other disasters.

Adequacy and sustainability of food availability and access

- 7. The concept of adequacy is particularly significant in relation to the right to food since it serves to underline a number of factors which must be taken into account in determining whether particular foods or diets that are accessible can be considered the most appropriate under given circumstances for the purposes of article 11 of the Covenant. The notion of sustainability is intrinsically linked to the notion of adequate food or food security, implying food being accessible for both present and future generations. The precise meaning of "adequacy" is to a large extent determined by prevailing social, economic, cultural, climatic, ecological and other conditions, while "sustainability" incorporates the notion of long-term availability and accessibility.
- 8. The Committee considers that the core content of the right to adequate food implies:

The availability of food in a quantity and quality sufficient to satisfy the dietary needs of individuals, free from adverse substances, and acceptable within a given culture;

The accessibility of such food in ways that are sustainable and that do not interfere with the enjoyment of other human rights.

- 9. Dietary needs implies that the diet as a whole contains a mix of nutrients for physical and mental growth, development and maintenance, and physical activity that are in compliance with human physiological needs at all stages throughout the life cycle and according to gender and occupation. Measures may therefore need to be taken to maintain, adapt or strengthen dietary diversity and appropriate consumption and feeding patterns, including breastfeeding, while ensuring that changes in availability and access to food supply as a minimum do not negatively affect dietary composition and intake.
- 10. Free from adverse substances sets requirements for food safety and for a range of protective measures by both public and private means to prevent contamination of foodstuffs through adulteration and/or through bad environmental hygiene or inappropriate handling at different stages throughout the food chain; care must also be taken to identify and avoid or destroy naturally occurring toxins.
- 11. Cultural or consumer acceptability implies the need also to take into account, as far as possible, perceived non-nutrient-based values attached to food and food consumption and informed consumer concerns regarding the nature of accessible food supplies.
- 12. Availability refers to the possibilities either for feeding oneself directly from productive land or other natural resources, or for well functioning distribution, processing and market systems that can move food from the site of production to where it is needed in accordance with demand.
- 13. Accessibility encompasses both economic and physical accessibility:

Economic accessibility implies that personal or household financial costs associated with the acquisition of food for an adequate diet should be at a level such that the attainment and satisfaction of other basic needs are not threatened or compromised. Economic accessibility applies to any acquisition pattern or entitlement through which people procure their food and is a measure of the extent to which it is satisfactory for the enjoyment of the right to adequate food. Socially vulnerable groups such as landless persons and other particularly impoverished segments of the population may need attention through special programmes.

Physical accessibility implies that adequate food must be accessible to everyone, including physically vulnerable individuals, such as infants and young children, elderly people, the physically disabled, the terminally ill and persons with persistent medical problems, including the mentally ill. Victims of natural disasters, people living in disaster-prone areas and other specially disadvantaged groups may need special attention and sometimes priority consideration with respect to

accessibility of food. A particular vulnerability is that of many indigenous population groups whose access to their ancestral lands may be threatened.

Obligations and violations

- 14. The nature of the legal obligations of States parties are set out in article 2 of the Covenant and has been dealt with in the Committee's General Comment No. 3 (1990). The principal obligation is to take steps to achieve progressively the full realization of the right to adequate food. This imposes an obligation to move as expeditiously as possible towards that goal. Every State is obliged to ensure for everyone under its jurisdiction access to the minimum essential food which is sufficient, nutritionally adequate and safe, to ensure their freedom from hunger.
- The right to adequate food, like any other human right, imposes three types or levels of obligations on States parties: the obligations to respect, to protect and to fulfil. In turn, the obligation to fulfil incorporates both an obligation to facilitate and an obligation to provide. Originally three levels of obligations were proposed: to respect, protect and assist/fulfil. (See Right to adequate food as a human right, Study Series No. 1, New York, 1989 (United Nations publication, Sales No. E.89.XIV.2).) The intermediate level of "to facilitate" has been proposed as a Committee category, but the Committee decided to maintain the three levels of obligation. The obligation to respect existing access to adequate food requires States parties not to take any measures that result in preventing such access. The obligation to protect requires measures by the State to ensure that enterprises or individuals do not deprive individuals of their access to adequate food. obligation to fulfil (facilitate) means the State must pro-actively engage in activities intended to strengthen people's access to and utilization of resources and means to ensure their livelihood, including food security. Finally, whenever an individual or group is unable, for reasons beyond their control, to enjoy the right to adequate food by the means at their disposal, States have the obligation to fulfil (provide) that right directly. This obligation also applies for persons who are victims of natural or other disasters.
- 16. Some measures at these different levels of obligations of States parties are of a more immediate nature, while other measures are more of a long-term character, to achieve progressively the full realization of the right to food.
- 17. Violations of the Covenant occur when a State fails to ensure the satisfaction of, at the very least, the minimum essential level required to be free from hunger. In determining which actions or omissions amount to a violation of the right to food, it is important to distinguish the inability from the unwillingness of a State party to comply. Should a State party argue that resource constraints make it impossible to provide access to food for those who are unable by themselves to secure such access, the State has to demonstrate that every effort has been made to use all the resources at its disposal in an effort to satisfy, as a matter of priority, those minimum obligations. This follows from article 2.1 of the Covenant, which obliges a State party to take the necessary steps to the maximum of its available resources, as previously pointed out by the Committee in its General Comment No. 3, paragraph 10. A State claiming that it is unable to carry out its

obligation for reasons beyond its control therefore has the burden of proving that this is the case and that it has unsuccessfully sought to obtain international support to ensure the availability and accessibility of the necessary food.

- 18. Furthermore, any discrimination in access to food, as well as to means and entitlements for its procurement, on the grounds of race, colour, sex, language, age, religion, political or other opinion, national or social origin, property, birth or other status with the purpose or effect of nullifying or impairing the equal enjoyment or exercise of economic, social and cultural rights constitutes a violation of the Covenant.
- 19. Violations of the right to food can occur through the direct action of States or other entities insufficiently regulated by States. These include: the formal repeal or suspension of legislation necessary for the continued enjoyment of the right to food; denial of access to food to particular individuals or groups, whether the discrimination is based on legislation or is pro-active; the prevention of access to humanitarian food aid in internal conflicts or other emergency situations; adoption of legislation or policies which are manifestly incompatible with pre-existing legal obligations relating to the right to food; and failure to regulate activities of individuals or groups so as to prevent them from violating the right to food of others, or the failure of a State to take into account its international legal obligations regarding the right to food when entering into agreements with other States or with international organizations.
- 20. While only States are parties to the Covenant and are thus ultimately accountable for compliance with it, all members of society individuals, families, local communities, non-governmental organizations, civil society organizations, as well as the private business sector have responsibilities in the realization of the right to adequate food. The State should provide an environment that facilitates implementation of these responsibilities. The private business sector national and transnational should pursue its activities within the framework of a code of conduct conducive to respect of the right to adequate food, agreed upon jointly with the Government and civil society.

Implementation at the national level

- 21. The most appropriate ways and means of implementing the right to adequate food will inevitably vary significantly from one State party to another. Every State will have a margin of discretion in choosing its own approaches, but the Covenant clearly requires that each State party take whatever steps are necessary to ensure that everyone is free from hunger and as soon as possible can enjoy the right to adequate food. This will require the adoption of a national strategy to ensure food and nutrition security for all, based on human rights principles that define the objectives, and the formulation of policies and corresponding benchmarks. It should also identify the resources available to meet the objectives and the most cost-effective way of using them.
- 22. The strategy should be based on a systematic identification of policy measures and activities relevant to the situation and context, as derived from the normative content of the right to adequate food and spelled out in

relation to the levels and nature of States parties' obligations referred to in paragraph 15 of the present General Comment. This will facilitate coordination between ministries and regional and local authorities and ensure that related policies and administrative decisions are in compliance with the obligations under article 11 of the Covenant.

- 23. The formulation and implementation of national strategies for the right to food requires full compliance with the principles of accountability, transparency, people's participation, decentralization, legislative capacity and the independence of the judiciary. Good governance is essential to the realization of all human rights, including the elimination of poverty and ensuring a satisfactory livelihood for all.
- 24. Appropriate institutional mechanisms should be devised to secure a representative process towards the formulation of a strategy, drawing on all available domestic expertise relevant to food and nutrition. The strategy should set out the responsibilities and time-frame for the implementation of the necessary measures.
- 25. The strategy should address critical issues and measures in regard to all aspects of the food system, including the production, processing, distribution, marketing and consumption of safe food, as well as parallel measures in the fields of health, education, employment and social security. Care should be taken to ensure the most sustainable management and use of natural and other resources for food at the national, regional, local and household levels.
- 26. The strategy should give particular attention to the need to prevent discrimination in access to food or resources for food. This should include: guarantees of full and equal access to economic resources, particularly for women, including the right to inheritance and the ownership of land and other property, credit, natural resources and appropriate technology; measures to respect and protect self-employment and work which provides a remuneration ensuring a decent living for wage earners and their families (as stipulated in article 7 (a) (ii) of the Covenant); maintaining registries on rights in land (including forests).
- 27. As part of their obligations to protect people's resource base for food, States parties should take appropriate steps to ensure that activities of the private business sector and civil society are in conformity with the right to food.
- 28. Even where a State faces severe resource constraints, whether caused by a process of economic adjustment, economic recession, climatic conditions or other factors, measures should be undertaken to ensure that the right to adequate food is especially fulfilled for vulnerable population groups and individuals.

Benchmarks and framework legislation

29. In implementing the country-specific strategies referred to above, States should set verifiable benchmarks for subsequent national and international monitoring. In this connection, States should consider the adoption of a framework law as a major instrument in the implementation of the

national strategy concerning the right to food. The framework law should include provisions on its purpose; the targets or goals to be achieved and the time-frame to be set for the achievement of those targets; the means by which the purpose could be achieved described in broad terms, in particular the intended collaboration with civil society and the private sector and with international organizations; institutional responsibility for the process; and the national mechanisms for its monitoring, as well as possible recourse procedures. In developing the benchmarks and framework legislation, States parties should actively involve civil society organizations.

30. Appropriate United Nations programmes and agencies should assist, upon request, in drafting the framework legislation and in reviewing the sectoral legislation. FAO, for example, has considerable expertise and accumulated knowledge concerning legislation in the field of food and agriculture. The United Nations Children's Fund (UNICEF) has equivalent expertise concerning legislation with regard to the right to adequate food for infants and young children through maternal and child protection including legislation to enable breastfeeding, and with regard to the regulation of marketing of breast milk substitutes.

Monitoring

31. States parties shall develop and maintain mechanisms to monitor progress towards the realization of the right to adequate food for all, to identify the factors and difficulties affecting the degree of implementation of their obligations, and to facilitate the adoption of corrective legislation and administrative measures, including measures to implement their obligations under articles 2.1 and 23 of the Covenant.

Remedies and accountability

- 32. Any person or group who is a victim of a violation of the right to adequate food should have access to effective judicial or other appropriate remedies at both national and international levels. All victims of such violations are entitled to adequate reparation, which may take the form of restitution, compensation, satisfaction or guarantees of non-repetition. National Ombudsmen and human rights commissions should address violations of the right to food.
- 33. The incorporation in the domestic legal order of international instruments recognizing the right to food, or recognition of their applicability, can significantly enhance the scope and effectiveness of remedial measures and should be encouraged in all cases. Courts would then be empowered to adjudicate violations of the core content of the right to food by direct reference to obligations under the Covenant.
- 34. Judges and other members of the legal profession are invited to pay greater attention to violations of the right to food in the exercise of their functions.
- 35. States parties should respect and protect the work of human rights advocates and other members of civil society who assist vulnerable groups in the realization of their right to adequate food.

<u>International obligations</u>

States parties

- 36. In the spirit of article 56 of the Charter of the United Nations, the specific provisions contained in articles 11, 2.1, and 23 of the Covenant and the Rome Declaration of the World Food Summit, States parties should recognize the essential role of international cooperation and comply with their commitment to take joint and separate action to achieve the full realization of the right to adequate food. In implementing this commitment, States parties should take steps to respect the enjoyment of the right to food in other countries, to protect that right, to facilitate access to food and to provide the necessary aid when required. States parties should, in international agreements whenever relevant, ensure that the right to adequate food is given due attention and consider the development of further international legal instruments to that end.
- 37. States parties should refrain at all times from food embargoes or similar measures which endanger conditions for food production and access to food in other countries. Food should never be used as an instrument of political and economic pressure. In this regard, the Committee recalls its position, stated in its General Comment No. 8, on the relationship between economic sanctions and respect for economic, social and cultural rights.

States and international organizations

- 38. States have a joint and individual responsibility, in accordance with the Charter of the United Nations, to cooperate in providing disaster relief and humanitarian assistance in times of emergency, including assistance to refugees and internally displaced persons. Each State should contribute to this task in accordance with its ability. The role of the World Food Programme (WFP) and the Office of the United Nations High Commissioner for Refugees (UNHCR), and increasingly that of UNICEF and FAO is of particular importance in this respect and should be strengthened. Priority in food aid should be given to the most vulnerable populations.
- 39. Food aid should, as far as possible, be provided in ways which do not adversely affect local producers and local markets, and should be organized in ways that facilitate the return to food self-reliance of the beneficiaries. Such aid should be based on the needs of the intended beneficiaries. Products included in international food trade or aid programmes must be safe and culturally acceptable to the recipient population.

The United Nations and other international organizations

40. The role of the United Nations agencies, including through the United Nations Development Assistance Framework (UNDAF) at the country level, in promoting the realization of the right to food is of special importance. Coordinated efforts for the realization of the right to food should be maintained to enhance coherence and interaction among all the actors concerned, including the various components of civil society. The food organizations, FAO, WFP and the International Fund for Agricultural Development (IFAD) in conjunction with the United Nations Development Programme (UNDP), UNICEF, the World Bank and the regional development banks,

should cooperate more effectively, building on their respective expertise, on the implementation of the right to food at the national level, with due respect to their individual mandates.

41. The international financial institutions, notably the International Monetary Fund (IMF) and the World Bank, should pay greater attention to the protection of the right to food in their lending policies and credit agreements and in international measures to deal with the debt crisis. Care should be taken, in line with the Committee's General Comment No. 2, paragraph 9, in any structural adjustment programme to ensure that the right to food is protected.
